

Received February 16, 2021, accepted February 22, 2021, date of publication March 2, 2021, date of current version March 11, 2021. *Digital Object Identifier* 10.1109/ACCESS.2021.3063104

# Integration of D2D, Network Slicing, and MEC in 5G Cellular Networks: Survey and Challenges

LUBNA NADEEM<sup>®1</sup>, MUHAMMAD AWAIS AZAM<sup>®2</sup>, YASAR AMIN<sup>®1</sup>, (Senior Member, IEEE), MOHAMMED A. AL-GHAMDI<sup>®3</sup>, KOK KEONG CHAI<sup>4</sup>, MUHAMMAD FAISAL NADEEM KHAN<sup>®5</sup>, AND MUHAMMAD ADNAN KHAN<sup>®6</sup>

<sup>1</sup>Department of Telecommunication Engineering, University of Engineering and Technology, Taxila, Taxila 47050, Pakistan

<sup>2</sup>Technology and Innovation Research Group, School of Information Technology, Whitecliffe 6011, New Zealand

<sup>3</sup>Computer Science Department, Umm Al-Qura University, Mecca 74200, Saudi Arabia
<sup>4</sup>Department of Electronic Engineering and Computer Science, Queen Mary University of London, London E1 4NS, U.K.

<sup>5</sup>Department of Electrical Engineering, University of Engineering and Technology, Taxila, Taxila 47050, Pakistan

<sup>6</sup>Faculty of Computing, Riphah School of Computing & Innovation, Riphah International University, Lahore Campus, Lahore 54000, Pakistan

Corresponding author: Lubna Nadeem (lubna.nadeem@uettaxila.edu.pk)

This work was supported by the Data and Artificial Intelligence Scientific Chair at Umm Al-Qura University, Makkah City, Saudi Arabia.

**ABSTRACT** With the tremendous demand for connectivity anywhere and anytime, existing network architectures should be modified. To cope with the challenges that arise due to the increasing flood of devices/users and a diverse range of application requirements, new technologies and concepts must be integrated to enable their benefits. Service providers and business companies are looking for new areas of research to enhance overall system performance. This article gives a detailed survey about the recent 5G technologies, the solutions they provide, and the effect caused by their addition to current cellular networks. It is based on the three most important 5G concepts: Device to Device (D2D), Network Slicing (NS), and Mobile Edge Computing (MEC). This study proposes to design the future 5G networks by the integration of all three technologies. It is believed that spectrum efficiency, energy efficiency, and overall throughput will be greatly improved by using D2D. The system delay and computational load will be reduced as tasks will be handled by edge routers located at the base stations. Thus offloading the core network and the system capital expenses and operational expenses will be reduced significantly by slicing the network.

**INDEX TERMS** 5G cellular networks, D2D communication, mobile edge computing, network slicing.

#### I. INTRODUCTION

Next big thing in the future is 5G technology, which is expected to grow three-folds by 2025 with applications ranging from robots, health services automation, smart agriculture machinery, and industrial control systems [1]. For 2G it took nearly 20 years from its deployment in 1991 to peak in 2011, and 3G recently peaked in 2016, 15 years from its original deployment. 4G is expected to peak in mobile subscriptions by 2022, 12 years from its original deployment. However, what differentiates it from other mobile generations is its rate of growth, as the 2G and 3G network's rate of decline increases. By 2020, the 2G, 3G, and 4G distributions were roughly 35.5%, 37.5% and 25% respectively, and by 2024 most of the world is anticipated to shift to 5G, covering almost 60% of the network space [2]. This is due to the

The associate editor coordinating the review of this manuscript and approving it for publication was Prakasam Periasamy<sup>(D)</sup>.

increasing demand for faster data, enormous devices, and various global initiatives to increase connectivity around the planet. Another reason for the massive adoption of 4G will be its backward compatibility in the future to help transition users to 5G. New mobile chips will allow 4G, 5G, and Wi-Fi to work together in unison to provide consistent connectivity wherever you are. If for example, 5G beamforming loses lineof-sight connection with your device due to some unforeseen obstacle, your device will be able to switch to a 4G network and back again when a connection with the 5G small cell is re-established, without any noticeable drop in service [1]. As you can see, 4G, more specifically LTE-A Pro is an integral part of the seamless transition to 5G. With the improvements in bandwidth and energy consumption that LTE-A Pro introduces and 5G will expand upon, more wireless subscribers coming online, there will be more competitive rates for larger data plans [3]. These gigabit level speeds, low latencies, and low costs will open many use cases for 5G such

as home routers with comparable speeds to Wi-Fi, seamlessly integrated augmented reality devices, and much more. The massive increase in the adoption of new services like mobile data downloading, sharing, and streaming has resulted in the worldwide rapid growth of mobile subscriptions. It is expected that by 2026, the mobile subscription will reach 8.8 billion with approx. 91% of them being mobile broadband. At the end of 2020, the global smartphone subscription was almost 6.1 billion, which is expected to be 7.5 billion by 2026 [4]. The cloud computation and network virtualization market in 5G is forecasted to be the value of USD 7.56 billion by 2025 [5].

In the future, applications may have variable needs in terms of data rate, delay, and reliability which is not possible with the same physical network infrastructure. NS is the concept that will create separate dedicated slices on the cloud and will help the users to utilize the network based on their demands. Like the scenario in which the person using online gaming requires more speed and less delay than the person using Facebook or WhatsApp. Crowded events like in musical concerts or cricket matches, there are a lot of people in a particular area who are sharing their live video or data with heavy applications so, they will benefit from network slicing and 5G. The organizers of the event must pay for a slice of the network that will boost the network capacity and internet speed-enhancing audiences' online experience [3]. Global System of Mobile-Communications (GSMA), in their research [6] stated that 5G network slicing can unlock the business opportunities to the estimate of \$300 billion amount by the year 2025. Still, the important research question is how the network slicing will meet up the demands of various verticals across 5G systems.

MEC has emerged as a promising concept that overcomes the burden of central cloud servers. This involves providing network storage, computing, and resources capabilities at the mobile edge nodes. Due to closer deployment to the user, the latency can be reduced significantly. This will enable the benefits of high bandwidth, delay-sensitive, load sharing, socially aware, and caching applications. The core and backhaul network offloading is a very important feature in MEC [7]. Each base station is equipped with an edge server so that it can perform all the computing tasks locally for proximity services in end users. Due to the dynamic traffic management requirements, the European Telecommunications Standards Institute (ETSI) introduces the concept of network virtualization and cloud computing facilities [8]. However, in the future MEC and network virtualization integration is expected to deliver network capabilities more efficiently and rapidly.

D2D concept enables the direct connection between devices in a cellular network without involving the base station or central fixed/wireless infrastructure [9]. The communication can be between two devices or it can be among multiple devices where the role of D2D is as a relay to provide good signal strength to the users present at the edge of the network. The prospective benefits that can be achieved through D2D are better spectrum utility, energy efficiency, less delay, improving system throughput, and battery saving [10]. System overall coverage and capacity can be enhanced significantly resulting in good economic and business gains.

This detailed research study is based on the comprehensive overview of the aforementioned concepts related to 5G, the history of research work in these areas separately, and the aspect of different benefits achieved by the combination of these different technologies. Finally the discussion on challenges, possible solutions and future trends.

## **II. CONTRIBUTIONS AND SCOPE**

5G technology is believed to revolutionize the completely different era of network designs and interfaces with a diverse range of applications. The challenges and hurdles in shifting to new technologies remain open issues. Many researchers, different government organizations, and industrial collaborations are striving their best to make this possible. To completely understand the vision of 5G, its various new technologies need to be investigated in detail. Table 1 provides the summary and comparison of previous survey works done in this area and their short-comings compared to the proposed work.

#### TABLE 1. Detail of Surveys.

Ref.	D2D	NS	MEC	Model
[7]	No	No	Yes	5G
[11]	Yes	No	No	5G
[12]	Yes	No	No	5G
[13]	No	Yes	No	5G
[14]	No	Yes	Yes	5G
[15]	Yes	Yes	No	5G
[16]	Yes	No	No	General
[17]	No	Yes	No	5G
[18]	No	No	Yes	General
[19]	No	No	Yes	5G
[20]	No	No	Yes	General
[21]	No	No	Yes	5G
[22]	No	Yes	Yes	5G
[23]	No	Yes	Yes	General
[24]	Yes	No	Yes	5G
[25]	No	No	Yes	5G
[26]	No	Yes	No	General
[Proposed]	Yes	Yes	Yes	5G

Over the past few years, a lot of work has been done in the field of 5G cellular networks. The existing literature work can be broadly categorized as follow: (1) Conventional 5G cellular networks, (2) D2D based 5G cellular networks, (3) Network slicing based 5G cellular networks, (4) Mobile edge and cloud computing 5G cellular networks, (5) D2D and MEC based 5G cellular networks, (6) D2D based



FIGURE 1. Structure of Paper.

virtualized 5G cellular networks, (7) Network slicing and MEC based 5G cellular networks. Extensive studies and techniques were considered for improving their performances but these researches were limited in one way or the other. Some of them presented limited evaluation and standardization policies, lacking detailed explanations of current research works and state of the art challenges. Similarly, no concrete study proposed the aspect of combining the above-mentioned technologies in a single work.

To the best of the author's knowledge, there is no existing work that covers the integration of D2D, NS, and MEC with 5G cellular networks and investigates the effect of 5G network architecture transformation and benefits. The scope of this study is to design the future next generations cellular networks in such a way that they can support a massive number of devices with an enormous range of customized applications in the least possible time without overloading the infrastructure and core network. This will result in enjoying a real-time gaming experience, critical emergency services, high speed, reliable remote applications, and many more with reduced cost.

The combination of technologies like network virtualization and 5G permit to construct end-to-end network design by partitioning the shared physical infrastructure. Due to the expected increase in the number of connected devices, it becomes necessary to incorporate network slicing with edge computing and D2D to accommodate the demand of the devices. This will build new network designs and realizations in the existing 5G networks.

# **III. STRUCTURE AND ORGANIZATION**

Fig. 1 represents the overview of the areas discussed in this work. We have provided the comprehensive details on the research work done in three major 5G technologies i.e, D2D, NS and MEC. Then the we have proposed the future architecture, discussion, challenges and future recommendations by integrating all of them. Section IV gives the brief summary of mobile communication technologies (1G to 5G). Section V is related to 5G business and service requirements with all the features, 5G used cases, and research challenges. Section VI is about D2D communication technology, its performance characteristics, the hierarchy, configuration, classification, and application scenarios. The detailed research work done in D2D resource management and optimization. Section VII discusses the work done in 5G NS technology, business perspectives, solutions, phases, and different layers. The research work in resource management with 5G NS and with D2D using NS. Section VIII is on MEC and research done in D2D with MEC. Section IX provides the architecture of the proposed model and explanation of its each layer. Section X gives a detailed discussion about the integration of D2D, NS and MEC. Section XI is related to open research challenges and future trends. and finally Section XII is about the conclusion.



FIGURE 2. Timeline of Mobile communication generations [27].

# **IV. WIRELESS COMMUNICATION**

Around the world, telecommunication companies and service providers have been struggling to propose the latest solutions for meeting the increasing demands of mobile data from business users and household customers. The beginning of the mobile wireless industry can be dated back to 1970. The cellular system became popular even more than was expected in the 1960s and 1970s. By the 1990s, this idea experienced tremendous growth as the number of mobile subscribers increased, and by 2010, there were four times extra mobile subscribers than the landlines.

Wireless communication technology has emerged because it was difficult to install wired links over long distances between the transmitter and the receiver. In 1895, the Italian scientist named Marconi was the first one to transmit the wireless signals through radio waves at a distance of 3.2 Km by using Morse code [27]. It was a break-through in the history of science when researchers and scientists started using radio frequency waves for communication. They started developing new devices that do not involve wired links and should be capable of providing mobility. This instant, widespread triumph of wireless communication concepts resulted in the evolution and improvement of standards due to diverse range of applications [28]. Wireless technology is beneficial in many ways:

- 1) Greater Efficiency
- 2) Less Hardware
- 3) More Flexibility
- 4) Low Cost

#### A. MOBILE COMMUNICATION (1G TO 5G)

The evolution of different mobile communication generations from 1G to 5G with time are in Fig 2.

In 1970, Martin Cooper developed the 1G mobile phone that provided two-way communication [27]. This generation

used analog signals and the major disadvantage was they were not able to cover long-distance due to signal attenuation [29].

2G allowed voice calls, short text messages and emails with digital technology. An example of 2G was (GSM) group special module whose name changed to Global System for Mobile Communication later with the speed of 14.4kbps to 64kbps [30]. 2G provided low internet speed with limited services support.

To provide better data rate, 2.5G named as General Packet Radio Service (GPRS) and 2.75G called as Enhanced Data-rate for GSM Evolution (EDGE) came in [29]. For GPRS, the data rate was 56 to 64 kbps, and for EDGE, it was around 170 kbps.

3G first emerged in 2001. It provided us with a speed of almost 384 kbps. The first 3G network was introduced commercially by the end of 2001 with WCDMA [31]. 3G became so popular that the number of subscribers reaches 295 Million at the end of 2007. 3G not only allowed for calls and text messages but also provided access to high-speed internet ranging from 200kbps to a few megabits per second.

With the emergence of the diverse range of services required by various applications, the researchers in 2009 proposed the technique on the benefits of multiple services (both real-time and non-real-time) introduced in CDMA based 3G networks [32]. The work suggested user's versatile quality of service (QoS) needs and power consumption control using cooperative game theory methods and finds the user's degree of satisfaction. Such a framework helped to evaluate various 3G technologies like CDMA and WLAN, their power, resource allocation, and QoS expectations. 3G called as Universal Mobile Telecommunications System (UMTS) introduced video calling for the first time. Then, smartphones became famous worldwide, with multimedia, games, health services, and a whole new social media experience [31].

3G technology was upgraded for more speed by few enhancements. The first was 3.5G with a very high data

rate of 2Mbps and the second was 3.75G [28]. Later, it evolved into an improved 3.9G system known as Long Term Evolution (LTE). The drawbacks of 3G were expensive infrastructure, costly spectrum demand, and expensive 3G smartphones.

The wireless communication technology named in 4G is LTE and LTE advanced. They provided very high-speed internet ranging from some megabits per second to gigabits per second. It was started in September 2008 with enhanced support for multimedia applications. The most important benefit of this was its compatibility with the previous generations [33].

The Fifth generation (5G) of mobile communication enables faster access to the network with low latency and high reliability. In this way, the data reaches the server with less delay and making possible the connection of many devices to the mobile network. It can provide a very high data rate, minimum transmission delay, more capacity, and the flexibility to add new services and features. The most important among them will be 100 times better capacity and enhanced internet speed than 4G [31], [33]. For example, 3G would take almost 15 hours to download a video of one hour, 4G would take about 5 minutes, but with 5G, it will be only 2 seconds. The response time in 4G was 50 milliseconds, which is now 400 times less in 5G having the value of almost one millisecond quicker than the blink of a human eye. Reducing transmission delay is very critical. Autonomous cars require a constant data connection with high speed and minimum delay [34].

Fig. 3 shows the comparison of data rates for mobile communication generations.



FIGURE 3. Data Rate comparison of Mobile technologies.

#### V. 5G BUSINESS AND SERVICE REQUIREMENTS

As 5G is garnering more attention, service providers are scrambling to buy more spectrum space and beginning to roll out more infrastructure as fast as possible. Over the coming years, there will be improvements in latency and speed. Currently, the LTE-A speed average around 50 to 100 megabits per second mostly. By the near future, at a conservative rate of growth of small cell deployments, massive MIMO arrays, and other technologies, the speed should average in the range of 500 megabits per second to 1 gigabit per second with latencies to about 25-millisecond range [35]. For the average mobile data consumer, these speeds will be more than enough

for a very long time, and the bandwidth should be large enough to support all the new devices that come to market.

# A. IMPORTANT 5G USE CASES

5G requirements and standards were defined by the governing bodies ITU and 3GPP [36], [37]. 5G telecommunication networks will comply with user's various QoS requirements. There are three major 5G used cases, proposed by standard organizations [38] shown in Fig. 4.

- enhanced-Mobile-Broadband (eMBB)
- massive-Machine-Type-Communication (mMTC)
- ultra-Reliable-Low-Latency-Communication (URLLC)



FIGURE 4. 5G Use Case scenarios (IMT-2020 and beyond) [39].

## 1) eMBB (ENHANCED MOBILE Broadband)

This case provides a very high-speed data rate and capacity even at the cell edge with faster connections setup and coverage. Examples include high-quality video downloading, 3D gaming, live video streaming, and many other intensive image and data scenarios to provide the required performance like virtual reality and high accuracy medical applications [38]. Nokia Communication published a report which predicted that from 2020 to 2030, there might be a drastic increase in traffic requirements up to 10,000 times over mobile broadband applications.

## 2) mMTC (MASSIVE MACHINE TYPE Communication)

5G based IoT scenario with tens of billions of attached devices and sensors. They support a large number of small data packets-based applications like smart home automation, intelligent wearables such as fitness bands, smart appliances, and many more that will improve with the advancement of 5G with time.

3) uRLLC (ULTRA-RELIABLE LOW LATENCY Communication) For highly crucial applications that require rapid communications with minimum latency. Examples are remote surgery and autonomous vehicular communications like self-driving cars that respond promptly to avoid accidents and any casualties [39]. The data should be highly reliable for 5G industrial and medical applications using the emerging concepts of machine learning and artificial intelligence for analysis.

It is reported [40] that by 2022, out of the 29 billion connected devices, the majority of them will be machine-tomachine devices. 5G promises to provide users QoS requirements with various scenarios like high data rate, reliability, and minimum delay [41]. There are cases that require wide-area coverage, the network should provide users with the desired speed at cell edges. In the case, where a huge number of low-power devices require reliable connections, the networks must be capable of connecting millions of devices with low power constraint low cost per device. Very high reliability and low latency are needed for the efficient performance of real-time applications [39].

## **B. 5G KEY FEATURES**

There are various features of 5G that make it unique and widespread.

- Date Rate: upto 10Gbps
- Speed: 10 to 100x better
- Latency: Less than 1-millisecond
- Bandwidth: 1000x more
- Devices: upto 100x more
- Availability: 99.999%
- Coverage: 100%
- Energy Usage: 90% reduction
- Battery Life: upto 10 years

# VI. D2D COMMUNICATION TECHNOLOGY

In cellular networks, the base stations (BS) is involved in the connection set-up between two user equipment (UE). For example, UE sends its data to BS using uplink (UL) resources, and then the BS redirects the data to a corresponding receiver using downlink (DL) resources. D2D communication refers to a radio technology that allows devices to directly exchange data without the use of a BS or its core network [42]. Fig. 5 clearly explains the idea of a direct communication link.



#### FIGURE 5. Communication links.

The Third Generation Partnership Project (3GPP) has investigated the proximity service (ProSe) communication under the control of cellular networks. 3GPP group divided ProSe communication into two parts proximity discovery and direct D2D communication [43], [44]. Details of the conceptual framework for incorporating D2D under the existing networks were discussed in [45], [46], where authors presented the concepts of device discovery, mode switching, user scheduling, and resources allocation for D2D based cellular networks.

# A. PERFORMANCE METRICS FOR D2D COMMUNICATION

The D2D performance criterion can improve the overall network throughput by using efficient and intelligent resource optimization techniques. Most important among them are:

- Throughput
- Spectrum Efficiency
- Energy Efficiency
- Capacity
- Secrecy Rate
- Latency

# **B. HIERARCHY OF D2D AREAS**

Fig.6 depicts the complete picture of different areas of research and their flow in D2D based communication.

# C. CONFIGURATION OF D2D COMMUNICATION

There are three different ways for D2D configuration [47].

# 1) NETWORK CONTROLLED D2D COMMUNICATION

In this scenario, the BS fully controls D2D communication (e.g. control signal, resources management, and discovering/establishing the connection). The centralized control results in efficient interference management and resource allocation. However, this configuration also causes high signaling overhead, wherein the number of D2D becomes large, and spectral efficiency (SE) is reduced [48].

# 2) AUTONOMOUS D2D COMMUNICATION

This scenario is like cognitive radio in which BS has no control over D2D users. Instead, D2D users leverage empty holes in the spectrum and sense a surrounding environment for obtaining channel state information (CSI), interference, and cellular user information. Although this method can avoid signaling overhead and time delay, communication security can be a potential issue. This configuration also causes unstable communication due to a lack of control.

## 3) NETWORK ASSISTED D2D COMMUNICATION

In this scenario, the BS supports D2D communication by controlling the signal and discovering/establishing the connection. Then, D2D users communicate in a self-organizing way, which reduces signal overhead. This configuration has merits described in the first two approaches.

# D. CLASSIFICATION OF D2D BASED CELLULAR NETWORKS

D2D based communication systems can be categorized based on the frequency bands used in In-band D2D communication and Out-band D2D communication [48], [49].



FIGURE 6. Hierarchy of Research Flow in D2D Communication.

## 1) INBAND COMMUNICATION

D2D communication uses a cellular network licensed spectrum. Based on spectrum sharing methods, inband D2D communication is divided into overlay and underlay mode. In the overlay inband, D2D and cellular users are assigned orthogonal resources (e.g., time/frequency). Hence, there is no interference between D2D and cellular users. However, it is insufficient in terms of spectrum efficiency (SE). In underlay inband, D2D and cellular users share time/frequency resources. Therefore, resulting in co-channel interference and efficient interference management techniques are required. Also, this method increases SE [49].

# 2) OUTBAND COMMUNICATION

D2D communication exploits the unlicensed ISM band also called industrial, scientific, and medical. Out-band communication eliminates interference between the users due to using a separate band instead of reusing therefore, it requires an extra radio interface. This type of communication adapts to other wireless technologies transmitting in the unlicensed band like Bluetooth and Wi-Fi. Outband communication is further divided into two types autonomous and controlled D2D based communication [44], [49].

37596

*E. APPLICATION SCENARIOS FOR D2D COMMUNICATION* There are certain scenarios that require the exchange of data between close nodes (e.g., cars, UEs, and sensors, among others). 3GPP defines three main instances for D2D, like public safety, network offloading, and commercial/social services. Additional scenarios have also been presented in the literature below:

## 1) PUBLIC SAFETY

The 3GPP standard proposed D2D communication for supporting emergency services and for meeting public service requirements [43]. LTE serves as an attractive solution for safety organizations (e.g., police, fire, and rescue services) that are required to intervene in case the system collapse. Failure may be the result of natural disasters (e.g., earthquakes, tornados, and hurricanes). It may be due to high congestion in extreme traffic load for crowded events (e.g., World Cup, Olympics). Safety organizations can rely on D2D technology to communicate information for short link communication between first responders.

# 2) COMMERCIAL AND SOCIAL SERVICES

D2D communication can support social networking in preference of fixed wireless infrastructure for communicating community information. Not only it can reduce resource usage



FIGURE 7. Examples of D2D Application Scenarios.

and alleviate network load, but also used for a) local promotions or advertisement from stores and restaurants located within close proximity to users [44], b) for broadcast data around community transportation facilities (e.g., schedules of trains at the station or flights information at the airport). Mobile multi-player gaming can also leverage D2D communication for social purposes. Direct link communication could offer advantages for game applications in terms of high rate, battery consumption, and low latency.

## 3) CELLULAR OFFLOADING

D2D communication can also be a key component for offloading network traffic. Cellular communication, when switched to D2D mode, can improve the spectrum utility because of increased overall system data rates, throughput, and proximity gain [50].

## 4) VEHICULAR NETWORKS

Vehicular networks are designed for vehicle-to-vehicle (V2V) communication, which is basically another application of D2D communication where nodes are vehicles. D2D communication can be used to meet strict delay and reliability requirements for Intelligent Transportation Systems (ITS). They can be implemented into collision avoidance systems by communicating road conditions (e.g., accident and road work locations) from vehicle to vehicle [51], [52].

## 5) CONTENT MULTI-CASTING

Content multi-casting via D2D communication works in such a way that a user with high channel quality is responsible for retransmitting received data from BS to users with weak channel quality. Data re-transmission is accomplished through D2D links [53].

# F. RESOURCE MANAGEMENT (RM) IN D2D

The research work for resource management in D2D [9], [54]–[56] can be broadly categorized into different areas as shown in Fig.8. Like, [56] proposed a technique for joint D2D mode selection and proportional scheduling which confirmed the minimum interference and the required



FIGURE 8. Research work in D2D Resource Optimization.

QoS demands were fulfilled. Also, the work in [57], [58] developed a joint resource scheduling and power management scheme for underlay case that guaranteed to optimize the network throughput and fairness among the users. The research work on power/energy-efficient mechanisms for D2D based LTE-A cellular networks were explored in [59] for underlay communication.

Most existing work related to spectrum reuse either in uplink [60]–[63] or downlink [64]–[70] has investigated D2D under traditional cellular networks. Downlink reuse scheme is more complicated than uplink due to high interference generated by the BSs to D2D users. In downlink reuse, the interference caused to D2D users depends exclusively on transmit power of BSs and their distance. Thus, designing efficient resource allocation schemes that can mitigate interference among cellular and D2D users is required to enhance the network performance. The resource allocation schemes include the joint mode selection, optimum power control, and channel assignment schemes for evaluating the performance of D2D under HetNets. Table. 2 summarizes the research work as mode selection (MS), power allocation (PA), resource allocation (RA), and joint PA and RA optimization.

## 1) RM IN D2D USING CONVENTIONAL METHODS

Extensive research work has been done on designing and developing algorithms for resource allocation in D2D based cellular networks. Researchers have utilized several approaches (e.g., optimization theory, game theory, and graph theory) to optimize various aspects of network performance (e.g., spectral efficiency, energy efficiency, and latency) in the presence of D2D communication. Existing literature can be divided into centralized and distributed approaches. For centralized methods, [85]–[88], the base station controls all the tasks of resource allocation, interference, and channel monitoring information within the cell. But with the increasing number of devices, the base stations are suffering from computational overhead. To deal with this issue many distributed resource allocation algorithms [89]–[91] have been

TABLE 2. Resource Management in D2D underlay Cellular Networks.

Ref.	D2D Range	No. of D2D/RB	MS	RA	PA	Solution Domain	Remarks
[60]	10-100m	One pair	No	Yes	Yes	Iterative Algorithm	Maximize EA and RA of D2D
[64]	20m	One pair	No	Yes	Yes	Matching Algorithms	Maximize D2D Throughput
[65]	20 m	One pair	Yes	Yes	Yes	Game Theory	Maximize System and D2D Throughput
[66]	25m	One pair	No	Yes	No	Heuristic Algorithm	Minimize Downlink Transmission Power
[67]	20m	One pair	No	Yes	No	Optimization	Maximize Overall and D2D rate
[68]	5m	Multiple pairs	No	Yes	No	Game Theory	Maximize Overall System Data-rate
[69]	10-20m	One pair	No	Yes	Yes	Auction Theory	Maximize System Data-rate
[70]	25m	Multiple pairs	No	Yes	No	Iterative Algorithm	Maximize Overall Network Rate
[71]	20m	One pair	No	Yes	Yes	Han-Kobayashi Algorithm	Maximize the Sum-rate
[72]	20m	One pair	No	Yes	Yes	Stable matching algorithm	Maximize EA and PA
[73]	20-50m	Multiple pairs	Yes	Yes	No	Iterative Optimization	Maximize D2D Sum-rate
[74]	20m	Multiple pairs	No	Yes	Yes	Graph Interference Model	Maximize Overall Throughput
[75]	20m	Multiple pairs	No	Yes	Yes	Robust Optimization Model	Maximize D2D users Sum-rate
[76]	50m	One pair	No	Yes	Yes	Convex Optimization	Maximize Energy Efficiency of D2D
[77]	15m	One pair	No	Yes	No	Analytic Solution	Maximize System Capacity
[78]	20-120m	Multiple pairs	No	Yes	Yes	Dinkelbach Approach	Optimize RA and PA of D2D
[79]	20m	One pair	No	Yes	Yes	Outer-approx Algorithm	Maximize System Energy Efficiency
[80]	25m	One pair	Yes	Yes	Yes	Non-Cooperative Game theory	Optimize Energy Efficiency of D2D
[81]	20-140m	One pair	No	Yes	No	Vertex search	Maximize Overall Throughput
[82]	20m	One pair	No	Yes	No	Greedy Heuristic	Maximize Network Throughput
[83]	15m	One pair	No	Yes	No	Matching-Hungarian Algorithm	Maximize System Throughput
[84]	30m	Multiple pairs	No	Yes	Yes	Non- Cooperative Game	Minimize System Interference

proposed in which the cellular and D2D users independently and opportunistically utilize the resources without involving base station in global information calculation. These schemes have the advantage in big network designs where interference mitigation is the concern.

The authors in [58], [92], investigated the method for improving the coverage area and maximizing throughput. [93] presented the work on energy efficiency optimization considering interference. The research in [42], [94] analyzed the initial step in D2D communication i.e, peer to peer discovery and device relaying for improving spectrum efficiency. For efficient resource optimization, the technique in [95], [96] proposed the multi-hop scenario. In [97], the authors designed the method for analyzing the performance based on D2D transmission time for improving throughput. The work proposed in [98] introduced a scheme for energy-efficient resource management in machine-to-machine (M2M) communication systems to deal with the limited spectrum and battery issues. They designed the matching algorithm-based technique to jointly optimize the device discovery, channel, power, and time slot allocation. It will maximize the energy efficiency of the overall system.

# 2) RM IN D2D USING MACHINE LEARNING

The conventional resource optimization schemes are not able to handle the increasing number of users with their variable channel state information (CSI). The resource management problems are mostly based on combinatorial models having nonlinear constraints for optimization which are difficult to solve. To meet up with the huge gap between the available communications technologies and future applications demands, smart and intelligent optimization schemes/algorithms must be introduced. Machine learning (ML), deep learning (DL), and reinforcement learning (RL) are the major areas of investigation in current research work that deals with the study of computational learning and pattern recognition in artificial intelligence to make decisions on complex scenarios and algorithms. This will help the cellular networks to maintain good QoS in terms of resource allocation (spectrum, power) and interference management. Further, the literature proposed in [99] provides an in-depth understanding of high-quality schemes using intelligent machine learning algorithms in communication systems [100]–[103]. Specifically, [103] proposed an intelligent resource management scheme for service provider selection and efficient power allocation in a wireless communication system. The scheme is based on the degree of mobile customers' satisfaction with QoS, considering the pricing policy and reputation of the service provider. For efficient resource management in D2D, ML plays an important role.

Works in literature focused on these solutions like in [104], the researchers introduced the method for providing ultra-reliable and low-latency services which are in high demand for future networks. The scheme is based on deep RL methods for channel and power allocation in D2D based system. A lot of work has been done in the recent past to solve the resource management problems in D2D using intelligent decision making and RL methods [105]-[108]. The work in [106] used the Q learning to maximize the system throughput, [107] used distributed method for spectrum allocation considering interference to maximize the network throughput, [108] proposed the intelligent resource allocation in D2D based vehicular networks using RL technique called as transfer actor-critic (AC). Currently, the concept of deep learning has been investigated for resource allocation scenarios [109]-[112]. The above-mentioned works in RL used the Markov decision process (MDP), which is useful when various D2D pairs or agents are renewing their policies at distinct times. But, when more than one D2D pairs or agents are revising their policies at the same time, then it will be a multi-agent problem. Recent works on multi-agent RL for resource optimization are discussed in [113]-[117]. The work done in [104], [117] is the most recent work in this regard for efficient resource allocation in D2D based systems using the latest machine learning algorithms to achieve desirable results.

## **VII. NETWORK SLICING TECHNOLOGY**

NS is a novel technology that permits innovative models for corporate sector and fulfills the increasing demands of future cellular networks such as:

- Increased network size.
- Increased traffic volume.
- Constantly changing networks.
- Increased network complexity.

This will enable the network operators to slice the single conventional network customized for specific services into multiple logical networks [13].

The goal is to modify the existing networks according to the applications demand and migrating from "a single size fits all" to the logical solution to create the slices with resources, isolation, and applications [118]. The NS can be done in numerous domains like radio access network, transport network or core network using cloud computing infrastructure or any other network functions and resources.

# A. 5G NS SOLUTIONS

The network slices provide an innovation sandbox that gives a level of isolation allowing mobile network operators the ability to build and deploy solutions without impacting all the network functions (NF) [7]. With this level of 5G automation and network slicing, service providers are expecting faster delivery of customized service level agreement (SLA) to increase the revenues and enabling new user experiences like:

- Slice life-cycle automation through simplified operations and reducing costs.
- Reduce time to market for delivering new 5G network services.
- Efficient and optimized use of valuable network resources with better insights and planning.
- The use of intelligent automation to gain an advantage in evolving and highly competitive landscape.
- To quickly meet customer's demands and ensuring their satisfaction.
- Increasing revenue with a reliable and scalable 5G network.

## B. 5G NETWORK VIRTUALIZATION (NV)

5G Network virtualization is an emerging method that utilizes software-based programming to create, execute, install, and control network elements and services [119]. NV intends to provide low cost and rapid response solution to 5G services with end-to-end supervision for improving customer's Quality of Experience (QoE) [120]. Its important realization is in applications like 5G end to end program integration and Network slice as a service [121]. 5G NV concepts were proposed and implemented in previous works as software defined networks (SDN) and network function virtualization (NFV). NV allows the physical network infrastructure to decouple its applications and services. Network hyper-visor is the main component that achieves NV, examine virtual networks, and assign virtual resources to them [122].

# C. 5G NS PHASES AND LAYERS

5G-based network slice comprises of different network functions for a particular business model or used case. The network slicing phases include preparation, instantiation, configuration, activation, runtime, and the decommissioning phase [119], [123]. Slicing is a new operating model accelerating the need for automation through cloud computing. NS has three types of layers [124] as in Fig.9:

#### 1) SERVICE INSTANCE LAYER

This layer supports the customer and business services with every single service representing a service instance for use.

#### 2) SLICE INSTANCE LAYER

This layer represents the instances that are available at the network. Each network slice instance delivers the system characteristics as desired by the upper service instance.

Ref.	Network	System Model	Objective	Remarks
[125]	Access	Business Model	Maximize data-rate	Provide more gains and fairness with dynamic slicing than static network slicing
[126]	Core + Access	Generalized Model	Minimize packet delay	Efficient resource allocation by virtual networks for vari- ous network topologies
[127]	Access	Business Model	Maximize revenue	Provide best pricing decisions on handling network slice request
[128]	Core	Economic Model	Minimize cost	Optimize virtual networks on various cloud networks for partnership
[129]	Access	Generalized Model	Maximize throughput	Efficient channel allocation with users association and power allocation method
[130]	Core	Schwarz-Christoffel	Minimize packet delay	Optimize the arrival rate of session requests
[131]	Access	Predictive Model	Maximize resource utilization	Predicts the trend and seasonal variation in future data traffic
[132]	Access	Predictive Model	Maximize energy efficiency	Efficient isolation in network slices with customized facil- ities in every slice
[133]	Core	Recovery Model	Maximize capacity	effects of random traffic variation with limited resources and robust
[134]	Core	Robust Model	Optimize bandwidth usage	Re-design and reroute the newly created VNFs due to failure
[135]	Core + Access	Generalized Model	Optimize bandwidth and time dependent pricing	To solve asymmetric information and price discrimination for software defined networks

TABLE 3.	Resource Management in Network slicing based 5G cellular Networks.
----------	--



FIGURE 9. Slice Layers Function.

## 3) RESOURCE INSTANCE LAYER

This layer offers entire system resources including both physical or virtual and different network functions required to generate a network slice instance.

## D. RESOURCE MANAGEMENT IN 5G NS

Table. 3 provides the overview of research work done using the NS concept in 5G cellular networks. It summarizes the comparison of system models considered in [125]–[135], the main objective of work and network models used.

## E. BUSINESS PERSPECTIVE OF 5G NS

5G NS is an emerging technique that provides a customized service experience to subscribers, using the concepts of virtualization and cloud computing. Today the market is driving the need for network slicing due to increased network size, traffic volumes, varying deployment models through hybrid cloud infrastructure [124]. All of them contribute to increased network complexity. Service providers are looking to build network slices based on the resources and components that are available in existing networks.

Depending on the vertical 5G used cases as discussed in Section V-A such as mMTC or eMBB, the network resources attached to those services are assigned that slice [124]. Network slicing uses the same physical infrastructure and creates logical partitions of the network based on the different SLA of the industries or vectors. Based on the 5G used cases and an increase in the no. of devices and services requires the need for slice automation. 5G automation is a key driver for network slicing which enables mobile network operators to build customized solutions and offerings.

## F. RESEARCH WORK IN D2D WITH NS

The research work on using D2D with network slicing concept in 5G cellular networks has not been explicitly explored to date. Previously, work has been done using SDN and virtualized networks which is still in its infancy [143]. For virtualized networks, the SDN controller or slice controller is responsible to direct the D2D connection for a successful session. This is a more effective method to coordinate D2D pairing or to use the existing cellular mode. The virtual SDN/slice controller directly sends commands to cellular devices with equivalent functions, despite using the control

TABLE 4.	Resource Manag	ement in Virtualized	l D2D based	Cellular Networks.
----------	----------------	----------------------	-------------	--------------------

Ref.	Network Model	System Model	Algorithm	Objective	Remarks
[136]	Traditional	Underlay/ Uplink	Heuristic	To optimize resource slicing of D2D	Increased system throughput with mini- mum interference
[137]	Virtualized	Underlay/ Uplink	Stochastic	To formulate the resource sharing problem for D2D in imperfect network state	Maximize the sum rate utility in imper- fect channel conditions
[138]	Traditional	Overlay	Stochastic	To allocate spectrum to D2D users in a multi- operator scenario	Efficient spectrum allocation scheme for considerable performance improvement
[139]	Virtualized	Underlay/ Uplink	Heuristic	To design a resource sharing scheme for D2D users in different virtual operators	Maximize Sum-Rate to achieve good performance
[140]	Virtualized	Underlay/ Downlink	Heuristic	To assign the virtual resources in D2D based LTE system considering interference	Maximize the overall system Sum rate
[141]	Virtualized	Underlay/ Downlink	Distributed	To design the information centric and virtu- alized D2D among different operators	Optimize the total system utility by effi- cient resource allocation
[142]	Virtualized	Underlay/ Uplink	Heuristic	To optimize the operator's profit by efficient resource allocation	Optimize the D2D and cellular users sum rate with maximizing the total profit

plane procedures. The virtualized D2D cellular networks must guarantee higher system performance and massive connectivity for devices to minimize the delay and hardware expenses as compared to current wireless networks. The research work in [144] proposed maximization of throughput for both D2D and cellular users in a virtualized LTE system. [140] verified that using virtualization for resource optimization can improve the system performance and throughput. Further, [139] proved better capacity and optimizing the sum rate when D2D users are related to multiple operators. Reference [143] studies the effect of resource aggregation in a virtualized common pool for D2D receivers and transmitters so that efficient resources can be allocated for available for D2D systems. Table. 4 presents the comparison of work done in D2D based cellular networks using virtualized networks. No detailed work is done in D2D with network slicing. This work is still in its initial phase.

# G. 5G NS FEATURES

Main features of 5G Network Slicing and virtualization are [145], [146]:

## 1) NETWORK FUNCTIONS AUTOMATION

This includes automation of dynamic network slice life-cycle (e.g., creating, adding, or deleting), resource optimization of networks, and dynamic interchange among data and administration planes [146].

## 2) IMPROVED SCALABILITY, ISOLATION AND RELIABILITY

These are the key aspects of network slicing in 5G that are capable of instant damage discovery mechanisms for every tenant and guarantees their security and dynamic performance needs [145]

## 3) PROGRAMMABILITY

This feature automates the availability of customer services and network management challenges particularly in communication applications [147]. The Programming of slices enables third parties to monitor and manage the assigned resources (like network and cloud resources). This helps in demand-oriented customized services and resource flexibility for virtual networks [148].

#### 4) HIERARCHICAL LEVEL ABSTRACTION

Network virtualization defines an abstraction layer by logically and physically partitioning different resources and network functions [149]. This enables on-demand service provisioning from a network slice, for instance, service providers and operators can facilitate manufacturing organizations to exploit the network as a service [150].

## 5) CUSTOMIZED SLICES

This feature is recognized at every level in the virtualized network with SDN in control and data plane. By using the concepts of Artificial Intelligence, the network virtualization on the data plane will deliver personalized services with value-added packages [148].

## 6) **RESOURCE FLEXIBILITY**

This will enable the assigned resources to scale up and down effectively according to the system requirements. The resource elasticity guarantees the requested service level agreements of customers irrespective of their location [151].

## **VIII. MOBILE EDGE COMPUTING (MEC)**

Mobile Edge Computing (MEC) is the term derived from previous cloud/central computing concept. In MEC all the storage, computation tasks are performed at edge of the network using small edge servers. This eliminates the need for transmitting information to the core network for processing. There are two major advantages of using the concept of MEC in existing cellular networks.

- Offloading: Energy saving for devices by offloading the exhaustive computing assignments.
- Congestion: The processing is now at the neighboring base station instead of the central cloud; therefore, the core network congestion and end-to-end delays are reduced significantly.

Recently, a lot of research has been done for MEC based networks to improve the system performance in improving energy usage [152]–[154] and delay [155]–[157]. However, the inadequate computing resources at the base station are not always sufficient to meet the requirements of devices.

# A. MEC WITH D2D

D2D is known for its benefit in offloading the neighboring device burden and reducing the central network congestion, so this motivates the combination of D2D with MEC providing advantages of resource reuse gain, hop gain, and proximity gain. These benefits are only possible by wisely designing the network and addressing the main issues that will arise due to the addition of D2D. Numerous studies in the literature have examined the effect of adding D2D offloading concept with MEC [158]–[161]. The work in [158], [159] investigated how the system performance and energy efficiency can be improved with D2D. Reference [160] proposed an innovative model of improving energy efficiency. The system will be offloading the tasks using D2D mobile devices for sharing the computation and communication resources. Similarly, for optimized system burden offloading, authors in [161] introduced a strategy for D2D using the non-causal processor information. The aforementioned work with joint D2D and MEC focused on improving the energy efficiency and reducing the end user's latency, but [162] proposed the scheme for computational capacity maximization. The work done in [104] is the very latest work for D2D using the concepts of MEC and machine learning. It is based on providing ultra-reliable and low latency applications in upcoming networks. By using a double deep Q-learning approach channel and power allocation problem can be solved efficiently in D2D based underlay cellular networks considering interference. The edge devices (D2D TXs) establishes the power assignment for improving spectral efficiency with desirable cellular users QoS.

# IX. PROPOSED 5G NETWORK WITH D2D, NS AND MEC

The network design of the proposed model can be shown in Fig. 10. It has various network sections [163]:

- Core network (CN)
- Transport network (TN)
- Mobile edge computing network (MEC)

# 1) CORE NETWORKS (CN)

Just like the previous SDN/NFV techniques, the NS based CNs are anticipated to be implemented as VNFs that can

run over virtual machines with the edge/fog based networks [164]. Such virtualized functions can be installed at a variety of locations in a network considering explicit application demands. Core network will slice the single common physical infrastructure into multiple logical slices depending on the application requirements. Like as in Fig. 10 Smartphones, Video control, smart city, smart wearable, e-health, smart industry, 3D Video are the services. Depending on these services the network management and orchestration plane will create the slices like eMBB slice, mMTC slice and URLLc slice.

The network slice controller is responsible for instantiating and managing the slice based on attributes characterized by the network slice type. It will begin the instantiating appeal to configure, create, and initiate the slice. After activation, the slice is controlled and monitored to fulfill the desired Quality of Service (QoS) or Service Level Agreement (SLA) requirements. The orchestrator and optimizer will monitor the network slice in real-time to make modifications and to meet the SLA [123]. The slice which is not required is deactivated and the allocated resources will be released.

# 2) TRANSPORT NETWORK (TN)

In future transport networks must be executed to accommodate numerous applications and customized services. The layout of slicing based transport networks can be implemented with suitable interfaces using NS infrastructures. The control panel will be responsible for resource discovery and optimization management. In a virtualized TNs, the factors like mobility management and load balancing aspects can be effectively coordinated with the interaction of RAN [165].

# 3) MOBILE EDGE COMPUTING (MEC) NETWORK

This is basically the physical layer which intends to shift network functions, resources, and content close at the consumer end. This is achieved by expanding traditional data processing units towards the network edge. The programming-based platforms that monitor the information-centric networking (ICN), will execute the NS process in edge networks [123], [166]. Edge computing is an emerging technique for intelligent machine learning and context-aware applications in edge devices of 5G network [7]. The aim is to deliver the increased throughput and quality of experience (QoE) for next-generation mobile networks including 5G and beyond [167]. With content-aware caching solutions and social relationships aware networks in edge computing servers, the virtualization of mobile edge networks ensures to minimize the load at the core networks. This will improve the capacity in case of high volume data processing, enabling real-time applications and optimized usage of resources [168].

The substrate resources of infrastructure provider are sliced into multiple virtual network operators called MVNO. In this way, capital expenditures (CAPEX) and operation expenditures (OPEX) of the network will reduce significantly [169].



FIGURE 10. 5G Network Slicing Architecture.

Each MVNO operating above virtual networks can deliver user's services on demand, which enables the attraction of many new subscribers. InP's will substantially gain more profits by renting individual virtual slices to different MVNOs [170].

The network utilization is increased by maximizing network resource sharing. Mostly by reusing different slices and by allocating new resources to different slices. This provides the mobile network operator with increased agility and faster time to revenue by providing a multi-tenant network slicing solution [121]. The business benefits of network slicing allow mobile network operators, the ability to automate the creation of network slices in a matter of minutes by using the same physical infrastructure.

By adding the D2D and edge routers at the radio access network shown in Fig. 10, the computation tasks can be done more efficiently, and backhaul load can be reduced considerably. 5G network slicing enables mobile network operators to become flexible and dynamically adjust network slices based on utilization capacity and customer SLA requirements. It is believed that by considering all the constraints properly and introducing innovative optimization techniques, the proposed network architecture will meet the goals of future generations.

#### X. DISCUSSION ON INTEGRATING D2D, NS AND MEC

The next generations of mobile communications, 5G and beyond are believed to have four main characteristics:

- Firstly, they are ultra-dense networks with the enormous growth in the number of mobile devices and data requirements.
- Secondly, the QoS needs are robust to sustain extremely high-data-rate, ultra-low-latency, reliability, and throughput.
- Thirdly, the networks should support heterogeneous technologies for compatibility of various user's types (like smartphones, tablets), network types (like cloud and edge computing, IEEE 802.11, Internet of things), and QoS types (like different values of speed, delay, and throughput for specific applications), and

• Fourthly, the massive multiple-input multiple-output (MIMO) concept at the base stations to minimize the interference and permit the nodes to broadcast simultaneously at the nearby locations.

Next generations data rate demand in terms of delay can be categorized as:

- Intense real-time: is for some strict latency requirement applications (like online gaming, health monitoring, and air traffic control).
- Mild real-time: is for services that can endure some delay values (like control of traffic signals, multimedia data transmission, and reception).
- Non-real-time: is for applications that can tolerate the delays and are not time-dependent (like email, forum communication).

MEC is known for its benefits in data with real-time applications (pre-defined latency) requirements using edge servers at nearby locations to the users. Such a decentralized network design will reduce the latency as compared to central cloud computing and will serve the demands of extremely interactive services with high computation and QoS needs. MEC have multiple mini edge servers placed at the edge network which may compute data in a single or collaborative mode to the cloud/central server placed at the core network [171], [172]. They will meet the 5G demand for ultra-low latency and will reduce the energy consumption by approx. 40% due to five times less access to the central cloud [25], [173].

MEC technology with small computing servers located at the edge of networks take advantage of previous NFV and SDN concepts, augmenting it with new storage/computing resources and creating multiple network slices based on services. MEC provide advantages of high bandwidth, less delay, offloading, location and real-time awareness about radio access networks. It will enable the improved customer's experience with good network QoS, less data transmission cost and reduced congestion. NS based MEC servers will deliver storage, computing and bandwidth capability that is shared by multiple logical slices directly connected to the base stations, owned by the infrastructure provider [174]. This will eradicate the requirement of forwarding data to the core network as most of the computation is performed at edge nodes resulting in lower latency.

Although a lot of work has been done independently in these areas the recent advances in integrating them are mostly ignored in the literature. Combining D2D with SDN and NFV was proposed in [175] which shows that throughput of the system can be improved in imperfect channel conditions. The work in in [140] proposed that using D2D in wireless virtual networks can significantly improve the system throughput. Similarly, [176] proposed the structure for software based information centric cellular network with D2D to optimize the network utility function. The optimization problem was based on resource allocation and caching parameters.

Adding D2D technology in the networks can provide the benefits of (1) Reuse gain (same resources can be used by

both cellular and D2D users), (2) Proximity gain (based on the distance between D2D devices to provide increased bit rate, low power consumption, and less delay), and (3) Hop gain (D2D devices can use a single link instead of uplink and downlink). The integration of D2D with NS and MEC requires a substantial modification in both core and access networks to support the smart services in future innovative infrastructures.

Slice controller is the main entity responsible for the management and operation of the network infrastructure [177]. NS helps to provide abstraction and isolation of network resources. 1) effectively managing hardware resources, 2) introducing innovative functions and services to the marketplace, 3) easily upgrade and maintain the network, 4) minimizing CAPEX and OPEX, 6) encouraging a diverse environment. Despite these advantages, the operation and maintenance of virtual network functions (VNFs) are challenging considering the MEC platform. NS has been considered as improved technology to provide scalability and flexibility of virtual resources.

NS is an essential technology to manage the diversity of D2D based services. The advantage of heterogeneous network services is that the devices can connect to multiple network slices (for specific services) at the same time. The concept of virtualization in D2D can be extended to include different 5G use cases scenarios. An important case in this regard is mMTC in which the devices communicate with each other. Incorporating network slicing with D2D communication is believed to enhance system performance in various mMTC applications [176]. The resources in a specific slice are isolated from other slices that can provide the improved quality of service (QoS) demands and thus alleviating the interference. D2D communication improves system throughput by less bandwidth usage, minimum path loss, low power consumption, and decreased cost as compared to traditional cellular communication.

By efficiently optimizing spectrum and energy efficiency in the proposed network, it is expected that higher throughput with less transmission delay and energy savings can be achieved. This survey examines features of 5G cellular networks, Network slicing, D2D based cellular networks, and MEC individually and what are the benefits and trends achieved when all these technologies will combine in upcoming future generation networks. This will reduce computational burden and delay for real-time applications in complex future network architecture [23].

# XI. OPEN CHALLENGES AND FUTURE TRENDS

The new 5G network architectures will be more complex than the existing ones, so every issue needs to be tackled very carefully for the best results.

# A. CHALLENGES IN D2D COMMUNICATION

The addition of D2D communication in existing network designs introduces technical challenges. Some of them are discussed in [44], [48], [49]:

## 1) SYNCHRONIZATION WITH PEER DISCOVERY

Throughout the device discovery stage, UEs try to find possible candidate UEs located within a specified proximity to establish direct communication. Then, synchronization among UEs is leveraged for efficient use of the available spectrum and of the UE energy [45], [49], [178]. There are generally two approaches: direct and network-assisted device discovery. In direct device discovery, UEs periodically broadcast discovery beacon signals. Hence, UEs located nearby can identify their presence and determine whether setting up D2D communication is warranted; however, since there is no synchronization between nodes, and receiver nodes continue to listen for beacon signals all time, this results in a UE battery drain and an increase in energy consumption. Also, this approach is distributed and does not involve BS in the discovery process. Thus, illegal users might listen to data during D2D communication. Another shortcoming of this approach is unrestrained usage of the licensed band [44], [49]. In network-assisted discovery, UEs inform the BS regarding their intent to connect and then send beacon signals. BS exchanges some messages, including identity and link information between two UEs for initiating a D2D link. This approach is centralized, and UEs listen only when instructed by the BS. The result is less energy consumption; however, this method comes at the cost of larger overhead, and limitations in privacy and scalability [42], [179].

## 2) MODE SELECTION

Mode selection can be described as the process of switching between the D2D mode (DM) or cellular mode (CM) should be used. Further after mode selection, the device must choose that if the D2D link will reuse the cellular links resources (i.e., underlay) or will use direct link (i.e., overlay). Mode switching can be performed statically before the D2D link setup or dynamically as per time slot basis. This is an important decision because sometimes direct link quality could be worse than cellular link quality. Design issues related to model selection can be described as follows. First, at what timescale should mode selection be performed and, subsequently, what measurement control signals are required, noting that the timescale cannot be too coarse. To avoid signal overhead, measurements and required control signaling should be kept at a minimum. Second, measurements (e.g., Signal-to-Noise ratio (SNR), path loss, distance) should be used to decide the mode of the users [46], [85].

## 3) INTERFERENCE MANAGEMENT

The major problem in D2D based communication is the interference caused due to reuse of cellular channel resources. Efficient spectrum allocation and power management techniques can significantly mitigate interference and maximize network performance, and they typically occur simultaneously with mode selection. This issue is quite challenging; therefore, proper allocation of resources is necessary to maintain the required user's QoS. Two types of interference can

occur cross-tier interference that is between cellular users and D2D pairs by reusing the same resource and co-tier interference that is among two different D2D pairs sharing a similar channel. Various interference management techniques were proposed for D2D communication under conventional cellular networks for both UL and DL spectrum reuse [85], [178]. When D2D communication coexists with heterogeneous cellular networks (i.e., HetNets), interference management techniques become more complicated, as they take into consideration the dense deployments of small cells.

The details of research work in resource (channel and power) allocation considering interference are thoroughly explained in Section VI-F. Similarly, Table 2 summarizes the work done in Underlay D2D communication where interference is the main constraint considered in problem-solving. Most of the work done in D2D is on single-cell networks but [84] proposed a detailed technique for interference mitigation in multi-cell D2D based cellular networks. They introduced a two-step method, where the first performs resource block allocation using a non-cooperative game and the second one divides the total interference problem into multiple minimization problems for efficient power allocation. Similarly, [180] provides a comprehensive understanding of resource allocation in next-generation wireless networks where the intra-cell and inter-cell interference has a great impact in evaluating the performance of the overall system. Notably, the interference mitigation problem of HetNets with D2D communication is still an open area for investigation. However, by using efficient and intelligent techniques for interference mitigation, the achievable data rate and QoS of end-users can be improved significantly.

## B. CHALLENGES IN INTEGRATING D2D, NS, AND MEC

#### 1) DYNAMIC RESOURCE ALLOCATION SCHEMES

Radio resource management is the major challenge that still needs to be improved for 5G core and radio access network domains. Available research work on 5G network slicing lacks the issues of new modulation techniques, massive MIMO, power consumption, and energy efficiency. The majority of the literature focused on finding the sub-optimal solutions by using different optimization methods. Next is to control power consumption to avoid interference in heterogeneous 5G cellular networks [11], [181].

## 2) LIMITATIONS OF EXISTING DEVICES

Using MEC technology for future cellular networks helps in overcoming the problems of centralized networks. But, they are faced with the challenges of device capabilities in caching the content, capacity, battery consumption, and storage. Consequently, when to offload the content and how to choose the caching node must be significant factors to consider [23]. D2D with MEC has been extensively explored over the past few years by academicians and industrialists for its benefits in offloading and emerging spectrum efficiency improvement many research issues need to be addressed to date.

# 3) CHANNEL CHARACTERISTICS

The most critical issue that should be investigated is the virtualization of physical channel characteristics and to enable isolation among various slices. The logical channel characteristics must be known to the devices so that they can support the device discovery, proximity aware services (ProSe), content caching, and relaying features in D2D edge computing. This is challenging in designing the cellular communication to avoid overhead and congestion losses.

# 4) COMPATIBILITY ISSUES

Integrating various radio access technologies (RATs) and their compatibility is challenging and extensive research work should be done. Several issues like resource utilization, interference management, power optimization congestion, and collision must be investigated in the upcoming networks. Advanced modulation techniques must be considered in designing the networks.

# 5) EDGE DEVICE RESOURCES

The mobile operators may install edge servers at the base stations to avoid burden at the cloud server and to achieve prompt response for customized services. Still here, the edge device's battery-saving and efficient utilization of scarce resources remain the test. New opportunities are required to work on resource scheduling techniques and virtualization without compromising the system performance and gains.

# 6) MOBILITY MANAGEMENT

For allocation of virtual resources, the mobility management factor of devices needs to be investigated with controlled handover to improve the spectrum utility in 5G networks [182]. The major challenge in D2D communication is its short-range requirement for a reliable link. As the distance between two devices increases, the connection will break so a solution must be required. For example, in the vehicle-to-vehicle scenario, the vehicle moves from the area of one base station to the other connected to a different edge server. The computation tasks are assigned to the new edge server through handover and the delay in this process is reduced by edge computing technology [22]. However, research in this area is very less and challenging. New optimized techniques need to be introduced for best network performance.

# 7) NEW FREQUENCY BANDS

Analysis must be done on the correlation between the unlicensed spectrum and licensed spectrum to efficiently utilize the current and new frequency bands. This is particularly becoming necessary with emerging spectrum sharing and spectrum sensing methods. Dynamic bandwidth slicing and sharing issues considering interference must require intelligent techniques [183], [184], therefore research must be done on novel intelligent learning algorithms like deep reinforcement learning techniques, machine learning, neural network, and long and short-term memory.

# 8) HARDWARE MODIFICATION

The shift to network virtualization of core and access layers are still very slow because most of the processing hardware components are dedicated and centralized. With the evolution of cloud-based networks, the components have started getting distributed and virtualized. The storage capacity is the main issue for dependence on dedicated hardware systems [185].

# 9) NOVEL SLICE CREATION TECHNIQUES

The authors in [186] concluded that innovative slice creation techniques based on descriptive configurations, informationoriented, and customized service requirements must be investigated across each network layer by intelligent slice orchestration. Slice controller management in dealing with resource allocation issues of all slices and dedicated slices must be explored.

# 10) SECURITY

Another important factor is the security that must be considered to provide the users with trust. As new services are emerging and customers every form of data is shifting to cloud networks, the serious concern is to develop efficient and intelligent techniques for privacy and secrecy [187], [188]. This will build the user's satisfaction and motivation to use new services.

# 11) NEW BUSINESS MODELS

An important challenge is to introduce versatile business models for operators and service providers which maximizes their utility functions and percentage of benefits achieved to shift from a centralized system [14], [189]. Good pricing and revenue based schemes must be explored for meeting the challenges of existing market. Pricing criteria for D2D acting as a relay and information centric networks is a serious concern of the service providers. Further, considering the new business opportunities and economic factors with both the single and multiple operators and infrastructure providers point of view.

# C. FUTURE TRENDS

The proposed work can be extended in many ways with most important recommendations are:

# 1) HETEROGENEOUS INTERNET OF EVERYTHING (IoE)

In the future, the devices are intended to provide multiple smart services. The devices should have the capability to enable heterogeneous IoE services like very high-frequency communication, virtual and augmented reality services, fully automated systems for telemedicine and transportation.

# 2) LONG PACKETS FOR COMMUNICATION

5G is built with the short communication packets having ultra-reliable and low latency features, but in the future, like 6G various applications e.g., augmented reality and autonomous vehicles need long communication packets

having ultra-high data rate and ultra-high reliability properties. Extensive research work with efficient optimization models must be designed to meet these future requirements.

# 3) 3D CELLULAR NETWORKS

Several new types of use cases and service providers are anticipated like 3D cellular network service providers, haptics communication providers, new nano-IoT applications, telemedicine service providers. These service providers are specialized to provide a whole new range of services different from 5G networks.

# 4) DYNAMIC D2D NETWORK ARCHITECTURE

A novel and dynamic D2D network architecture with high mobility case must be designed for reliable and long-lasting connection that allows control of reliable connections. Efficient network models for D2D based heterogeneous networks scenarios are required for improving throughput and QoS requirements.

# 5) INNOVATIVE NETWORK SLICING ALGORITHMS

New innovative business models are required for enabling network slicing in future generations of mobile networks [14]. Adaptive schemes for network slicing life cycle management are expected to manage and control the dynamic features of future networks. The network slice creation, deletion, and update must be linked with the service time scale depending on the diverse QoS demands.

## 6) NETWORK SLICING FOR 3D NETWORKS

The future equipment and base stations are predicted to be drone-based like in 6G cellular networks. Network slicing-based 3D cellular networks are recommended in the future for efficient orchestration and management of network function resources.

## 7) INTELLIGENT MACHINE LEARNING MODELS

For future networks, machine learning is believed to be an essential part of managing smart applications. By using these concepts intelligent network slicing and edge computing models can be designed. Machine learning requires initial training data sets and detailed simulations for tuning the machine learning parameters. Deep Q learning and Reinforcement learning concepts are recommended for future generations of cellular networks due to the rapid increase in intelligent smart applications. New meta-learning based techniques should be introduced for network slicing and edge computing to minimize the delay in transmission.

# **XII. CONCLUSION**

The proposed work gives a comprehensive understanding about the recent hot topics in 5G technology i.e., D2D, NS, and MEC. Many schemes were previously proposed by the researchers but were incomplete in one way or the other. They were based on a trade-off between different parameters. D2D, MEC, and NS are continuously developing independently. The conventional cellular network design is not capable to support emerging applications demand. This study addresses the idea of integrating D2D, MEC, and NS to achieve their individual gains. Such a network design will open a new era of revenues for mobile operators while reducing the complexity and cost of the system. This will fill the gap with multiple domain-based systems to enable the availability of resources according to customized slice requirements by efficient management of slice controller. The benefit of MEC is also promising to add in current networks to deal with complex situations. NS has given rise to a plethora of emerging new vertical industries and services each requiring its specific network requirements. This has led to the increased complexity and the need to consolidate solutions into one vendor a multi-tenant network slicing solution enabling to unlock new revenue streams and services. By using D2D, the spectrum can be used more efficiently with less delay. Other improvements include a reduction in time to market and an increase in network efficiency as well as minimizing network complexity, with reduced latency, increased throughput, and increased connection density.

# REFERENCES

- G. A. Akpakwu, B. J. Silva, G. P. Hancke, and A. M. Abu-Mahfouz, "A survey on 5G networks for the Internet of Things: Communication technologies and challenges," *IEEE Access*, vol. 6, pp. 3619–3647, 2018.
- [2] A. Ghosh, A. Maeder, M. Baker, and D. Chandramouli, "5G evolution: A view on 5G cellular technology beyond 3GPP release 15," *IEEE Access*, vol. 7, pp. 127639–127651, 2019.
- [3] A. Agarwal, G. Misra, and K. Agarwal, "The 5th generation mobile wireless networks-key concepts, network architecture and challenges," *Amer. J. Electr. Electron. Eng.*, vol. 3, no. 2, pp. 22–28, 2015.
- [4] Ericsson. (Feb. 2020). Mobile Subscriptions Forecast–Mobility Report– Ericsson. [Online]. Available: https://www.ericsson.com/en/mobilityreport/
- [5] (Jan. 2019). 5G Technology Market Size, Share, Growth and Industry Analysis. [Online]. Available: https://www.marketresearchfuture. com/reports/5g-technology-market-2988
- [6] A. Blenk, A. Basta, M. Reisslein, and W. Kellerer, "Survey on network virtualization hypervisors for software defined networking," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 1, pp. 655–685, 1st Quart., 2016.
- [7] T. X. Tran, A. Hajisami, P. Pandey, and D. Pompili, "Collaborative mobile edge computing in 5G networks: New paradigms, scenarios, and challenges," *IEEE Commun. Mag.*, vol. 55, no. 4, pp. 54–61, Apr. 2017.
- [8] Mobile Edge Computing (MEC); Framework and Reference Architecture, M. ETSI, ETSI, DGS MEC, Sophia Antipolis, France, 2019, vol. 3.
- [9] L. Wei, R. Q. Hu, Y. Qian, and G. Wu, "Enable device-to-device communications underlaying cellular networks: Challenges and research aspects," *IEEE Commun. Mag.*, vol. 52, no. 6, pp. 90–96, Jun. 2014.
- [10] Z. Zhou, M. Dong, K. Ota, J. Wu, and T. Sato, "Energy efficiency and spectral efficiency tradeoff in device-to-device (D2D) communications," *IEEE Wireless Commun. Lett.*, vol. 3, no. 5, pp. 485–488, Oct. 2014.
- [11] M. Noura and R. Nordin, "A survey on interference management for device-to-device (D2D) communication and its challenges in 5G networks," J. Netw. Comput. Appl., vol. 71, pp. 130–150, Aug. 2016.
- [12] M. Noor-A-Rahim, Z. Liu, H. Lee, G. M. N. Ali, D. Pesch, and P. Xiao, "A survey on resource allocation in vehicular networks," *IEEE Trans. Intell. Transp. Syst.*, early access, Sep. 4, 2020, doi: 10.1109/TITS.2020.3019322.
- [13] H. Zhang, N. Liu, X. Chu, K. Long, A.-H. Aghvami, and V. C. Leung, "Network slicing based 5G and future mobile networks: Mobility, resource management, and challenges," *IEEE Commun. Mag.*, vol. 55, no. 8, pp. 138–145, Aug. 2017.
- [14] Y. Sun, M. Peng, S. Mao, and S. Yan, "Hierarchical radio resource allocation for network slicing in fog radio access networks," *IEEE Trans. Veh. Technol.*, vol. 68, no. 4, pp. 3866–3881, Apr. 2019.

- [15] K. Wang, F. R. Yu, and H. Li, "Information-centric virtualized cellular networks with device-to-device communications," *IEEE Trans. Veh. Technol.*, vol. 65, no. 11, pp. 9319–9329, Nov. 2016.
- [16] F. Jameel, Z. Hamid, F. Jabeen, S. Zeadally, and M. A. Javed, "A survey of device-to-device communications: Research issues and challenges," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 3, pp. 2133–2168, 3rd Quart., 2018.
- [17] E. J. Kitindi, S. Fu, Y. Jia, A. Kabir, and Y. Wang, "Wireless network virtualization with SDN and C-RAN for 5G networks: Requirements, opportunities, and challenges," *IEEE Access*, vol. 5, pp. 19099–19115, 2017.
- [18] Y. Mao, C. You, J. Zhang, K. Huang, and K. B. Letaief, "A survey on mobile edge computing: The communication perspective," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 4, pp. 2322–2358, 4th Quart., 2017.
- [19] F. Vhora and J. Gandhi, "A comprehensive survey on mobile edge computing: Challenges, tools, applications," in *Proc. 4th Int. Conf. Comput. Methodologies Commun. (ICCMC)*, Mar. 2020, pp. 49–55.
- [20] A. Shakarami, M. Ghobaei-Arani, M. Masdari, and M. Hosseinzadeh, "A survey on the computation offloading approaches in mobile edge/cloud computing environment: A stochastic-based perspective," J. Grid Comput., vol. 18, pp. 1–33, Aug. 2020.
- [21] L. Liu, C. Chen, Q. Pei, S. Maharjan, and Y. Zhang, "Vehicular edge computing and networking: A survey," *Mobile Netw. Appl.*, vol. 25, pp. 1–24, Jul. 2020.
- [22] S. A. Abdel Hakeem, A. A. Hady, and H. Kim, "5G-V2X: Standardization, architecture, use cases, network-slicing, and edge-computing," *Wireless Netw.*, vol. 26, no. 8, pp. 6015–6041, Nov. 2020.
- [23] L. Cominardi, T. Deiss, M. Filippou, V. Sciancalepore, F. Giust, and D. Sabella, "MEC support for network slicing: Status and limitations from a standardization viewpoint," *IEEE Commun. Standards Mag.*, vol. 4, no. 2, pp. 22–30, Jun. 2020.
- [24] R. Wang, J. Yan, D. Wu, H. Wang, and Q. Yang, "Knowledge-centric edge computing based on virtualized D2D communication systems," *IEEE Commun. Mag.*, vol. 56, no. 5, pp. 32–38, May 2018.
- [25] P. Porambage, J. Okwuibe, M. Liyanage, M. Ylianttila, and T. Taleb, "Survey on multi-access edge computing for Internet of Things realization," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 4, pp. 2961–2991, 4th Quart., 2018.
- [26] R. Su, D. Zhang, R. Venkatesan, Z. Gong, C. Li, F. Ding, F. Jiang, and Z. Zhu, "Resource allocation for network slicing in 5G telecommunication networks: A survey of principles and models," *IEEE Netw.*, vol. 33, no. 6, pp. 172–179, Nov. 2019.
- [27] M. Meraj and S. Kumar, "Evolution of mobile wireless technology from 0G to 5G," *Int. J. Comput. Sci. Inf. Technol.*, vol. 6, no. 3, pp. 2545–2551, 2015.
- [28] A. Osseiran, F. Boccardi, V. Braun, K. Kusume, P. Marsch, M. Maternia, O. Queseth, M. Schellmann, H. Schotten, H. Taoka, H. Tullberg, M. A. Uusitalo, B. Timus, and M. Fallgren, "Scenarios for 5G mobile and wireless communications: The vision of the METIS project," *IEEE Commun. Mag.*, vol. 52, no. 5, pp. 26–35, May 2014.
- [29] M. R. Bhalla and A. V. Bhalla, "Generations of mobile wireless technology: A survey," Int. J. Comput. Appl., vol. 5, no. 4, pp. 26–32, Aug. 2010.
- [30] G. S. Nitesh and A. Kakkar, "Generations of mobile communication," Int. J. Adv. Res. Comput. Sci. Softw. Eng., vol. 32, no. 3, pp. 320–324, 2016.
- [31] M. Ramzan and J. A. Shaheen, "Comparison: 3G wireless networks with 4G wireless networks technology wise," Int. J. Adv. Sci. Technol., vol. 108, pp. 1–10, Nov. 2017.
- [32] E. E. Tsiropoulou, T. Kastrinogiannis, and S. Papavassiliou, "Uplink power control in QoS-aware multi-service CDMA wireless networks," *J. Commun.*, vol. 4, no. 9, pp. 654–668, Oct. 2009.
- [33] E. Ezhilarasan and M. Dinakaran, "A review on mobile technologies: 3G, 4G and 5G," in Proc. 2nd Int. Conf. Recent Trends Challenges Comput. Models (ICRTCCM), Feb. 2017, pp. 369–373.
- [34] S. Yrjola, "Technology antecedents of the platform-based ecosystemic business models beyond 5G," in *Proc. IEEE Wireless Commun. Netw. Conf. Workshops (WCNCW)*, Apr. 2020, pp. 1–8.
- [35] A. Morgado, K. M. S. Huq, S. Mumtaz, and J. Rodriguez, "A survey of 5G technologies: Regulatory, standardization and industrial perspectives," *Digit. Commun. Netw.*, vol. 4, no. 2, pp. 87–97, Apr. 2018.
- [36] A. Hoglund, D. P. Van, T. Tirronen, O. Liberg, Y. Sui, and E. A. Yavuz, "3GPP release 15 early data transmission," *IEEE Commun. Standards Mag.*, vol. 2, no. 2, pp. 90–96, Jun. 2018.

- [37] [SPEC] 3GPP TR 38.913—Study on Scenarios and Requirements for Next Generation Access Technologies—iTecTec. Accessed: Feb. 3, 2021. [Online]. Available: https://itectec.com/archive/3gpp-specification-tr-38-913/
- [38] P. Popovski, K. F. Trillingsgaard, O. Simeone, and G. Durisi, "5G wireless network slicing for eMBB, URLLC, and mMTC: A communicationtheoretic view," *IEEE Access*, vol. 6, pp. 55765–55779, 2018.
- [39] (Sep. 2015). M.2083: IMT Vision–Framework and Overall Objectives of the Future Development of IMT for 2020 and Beyond. [Online]. Available: https://www.itu.int/rec/R-REC-M.2083-0-201509-I
- [40] W. Obile, "Ericsson mobility report," Ericsson, Stockholm, Sweden, Tech. Rep., Nov. 2016.
- [41] D. Warren and C. Dewar. (Dec. 2014). GSMA | Understanding 5G: Perspectives on Future Technological Advancements in Mobile–GSMAI Report–Future Networks. [Online]. Available: https://www.gsma.com/futurenetworks/resources/
- [42] G. Fodor, E. Dahlman, G. Mildh, S. Parkvall, N. Reider, G. Miklós, and Z. Turányi, "Design aspects of network assisted device-to-device communications," *IEEE Commun. Mag.*, vol. 50, no. 3, pp. 170–177, Mar. 2012.
- [43] A. Patil and H. Sawant, "Technical specification group services and system aspects, IP multimedia subsystem (IMS)," *Int. J. Electron. Commun. Comput. Eng.*, vol. 3, no. 2, pp. 234–238, 2012.
- [44] A. Pyattaev, O. Galinina, K. Johnsson, A. Surak, R. Florea, S. Andreev, and Y. Koucheryavy, "Network-assisted D2D over WiFi direct," in *Smart Device to Smart Device Communication*. Springer, 2014, pp. 165–218.
- [45] K. Doppler, M. P. Rinne, P. Janis, C. Ribeiro, and K. Hugl, "Deviceto-device communications; functional prospects for LTE-advanced networks," in *Proc. IEEE Int. Conf. Commun. Workshops*, Jun. 2009, pp. 1–6.
- [46] G. Fodor, S. Roger, N. Rajatheva, S. Ben Slimane, T. Svensson, P. Popovski, J. M. B. Da Silva, and S. Ali, "An overview of device-to-device communications technology components in METIS," *IEEE Access*, vol. 4, pp. 3288–3299, 2016.
- [47] A. Asadi, Q. Wang, and V. Mancuso, "A survey on device-to-device communication in cellular networks," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 4, pp. 1801–1819, 4th Quart., 2014.
- [48] L. Song, D. Niyato, Z. Han, and E. Hossain, Wireless Device-to-Device Communications and Networks. Cambridge, U.K.: Cambridge Univ. Press, 2015.
- [49] P. Mach, Z. Becvar, and T. Vanek, "In-band device-to-device communication in OFDMA cellular networks: A survey and challenges," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 1885–1922, 4th Quart., 2015.
- [50] S. Andreev, A. Pyattaev, K. Johnsson, O. Galinina, and Y. Koucheryavy, "Cellular traffic offloading onto network-assisted device-to-device connections," *IEEE Commun. Mag.*, vol. 52, no. 4, pp. 20–31, Apr. 2014.
- [51] X. Cheng, L. Yang, and X. Shen, "D2D for intelligent transportation systems: A feasibility study," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 4, pp. 1784–1793, Aug. 2015.
- [52] A. Algedir and H. H. Refai, "Energy-efficient D2D communication under downlink HetNets," in *Proc. IEEE Wireless Commun. Netw. Conf.* (WCNC), Apr. 2019, pp. 1–6.
- [53] Y. Zhang, F. Li, X. Ma, K. Wang, and X. Liu, "Cooperative energyefficient content dissemination using coalition formation game over device-to-device communications," *Can. J. Electr. Comput. Eng.*, vol. 39, no. 1, pp. 2–10, 2016.
- [54] M. Hasan, E. Hossain, and D. I. Kim, "Resource allocation under channel uncertainties for relay-aided device-to-device communication underlaying LTE—A cellular networks," *IEEE Trans. Wireless Commun.*, vol. 13, no. 4, pp. 2322–2338, Apr. 2014.
- [55] H. H. Esmat, M. M. Elmesalawy, and I. I. Ibrahim, "Joint channel selection and optimal power allocation for multi-cell D2D communications underlaying cellular networks," *IET Commun.*, vol. 11, no. 5, pp. 746–755, Mar. 2017.
- [56] S. Wen, X. Zhu, X. Zhang, and D. Yang, "QoS-aware mode selection and resource allocation scheme for device-to-device (D2D) communication in cellular networks," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC)*, Jun. 2013, pp. 101–105.
- [57] X. Li, R. Shankaran, M. A. Orgun, G. Fang, and Y. Xu, "Resource allocation for underlay D2D communication with proportional fairness," *IEEE Trans. Veh. Technol.*, vol. 67, no. 7, pp. 6244–6258, Jul. 2018.
- [58] C.-H. Yu, K. Doppler, C. B. Ribeiro, and O. Tirkkonen, "Resource sharing optimization for device-to-device communication underlaying cellular networks," *IEEE Trans. Wireless Commun.*, vol. 10, no. 8, pp. 2752–2763, Aug. 2011.

- [59] L. Xu, C. Jiang, Y. Shen, T. Q. S. Quek, Z. Han, and Y. Ren, "Energy efficient D2D communications: A perspective of mechanism design," *IEEE Trans. Wireless Commun.*, vol. 15, no. 11, pp. 7272–7285, Nov. 2016.
- [60] Z. Zhou, K. Ota, M. Dong, and C. Xu, "Energy-efficient matching for resource allocation in D2D enabled cellular networks," *IEEE Trans. Veh. Technol.*, vol. 66, no. 6, pp. 5256–5268, Jun. 2017.
- [61] Z. Zhou, M. Dong, K. Ota, G. Wang, and L. T. Yang, "Energy-efficient resource allocation for D2D communications underlaying cloud-RANbased LTE—A networks," *IEEE Internet Things J.*, vol. 3, no. 3, pp. 428–438, Jun. 2016.
- [62] R. Yin, C. Zhong, G. Yu, Z. Zhang, K. K. Wong, and X. Chen, "Joint spectrum and power allocation for D2D communications underlaying cellular networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 4, pp. 2182–2195, Apr. 2016.
- [63] Y. Jiang, Q. Liu, F. Zheng, X. Gao, and X. You, "Energy-efficient joint resource allocation and power control for D2D communications," *IEEE Trans. Veh. Technol.*, vol. 65, no. 8, pp. 6119–6127, Aug. 2016.
- [64] T. Huynh, T. Onuma, K. Kuroda, M. Hasegawa, and W.-J. Hwang, "Joint downlink and uplink interference management for device to device communication underlaying cellular networks," *IEEE Access*, vol. 4, pp. 4420–4430, 2016.
- [65] M. Khan, M. Alam, Y. Moullec, and E. Yaacoub, "Throughput-aware cooperative reinforcement learning for adaptive resource allocation in device-to-device communication," *Future Internet*, vol. 9, no. 4, p. 72, Nov. 2017.
- [66] X. Xiao, X. Tao, and J. Lu, "A QoS-aware power optimization scheme in OFDMA systems with integrated device-to-device (D2D) communications," in *Proc. IEEE Veh. Technol. Conf. (VTC Fall)*, Sep. 2011, pp. 1–5.
- [67] R. AliHemmati, B. Liang, M. Dong, G. Boudreau, and S. H. Seyedmehdi, "Power allocation for underlay device-to-device communication over multiple channels," *IEEE Trans. Signal Inf. Process. Over Netw.*, vol. 4, no. 3, pp. 467–480, Sep. 2018.
- [68] C. Xu, L. Song, Z. Han, Q. Zhao, X. Wang, and B. Jiao, "Interferenceaware resource allocation for device-to-device communications as an underlay using sequential second price auction," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2012, pp. 445–449.
- [69] C. Xu, L. Song, Z. Han, Q. Zhao, X. Wang, X. Cheng, and B. Jiao, "Efficiency resource allocation for device-to-device underlay communication systems: A reverse iterative combinatorial auction based approach," *IEEE J. Sel. Areas Commun.*, vol. 31, no. 9, pp. 348–358, Sep. 2013.
- [70] M. Hasan and E. Hossain, "Distributed resource allocation in D2Denabled multi-tier cellular networks: An auction approach," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2015, pp. 2949–2954.
- [71] X. Chen, L. Chen, M. Zeng, X. Zhang, and D. Yang, "Downlink resource allocation for device-to-device communication underlaying cellular networks," in *Proc. IEEE 23rd Int. Symp. Pers., Indoor Mobile Radio Commun. (PIMRC)*, Sep. 2012, pp. 232–237.
- [72] C.-H. Yu and O. Tirkkonen, "Device-to-device underlay cellular network based on rate splitting," in *Proc. IEEE Wireless Commun. Netw. Conf.* (WCNC), Apr. 2012, pp. 262–266.
- [73] D. Zhu, J. Wang, A. L. Swindlehurst, and C. Zhao, "Downlink resource reuse for device-to-device communications underlaying cellular networks," *IEEE Signal Process. Lett.*, vol. 21, no. 5, pp. 531–534, May 2014.
- [74] Z. Yang, N. Huang, H. Xu, Y. Pan, Y. Li, and M. Chen, "Downlink resource allocation and power control for device-to-device communication underlaying cellular networks," *IEEE Commun. Lett.*, vol. 20, no. 7, pp. 1449–1452, Jul. 2016.
- [75] R. Zhang, X. Cheng, L. Yang, and B. Jiao, "Interference graph-based resource allocation (InGRA) for D2D communications underlaying cellular networks," *IEEE Trans. Veh. Technol.*, vol. 64, no. 8, pp. 3844–3850, Aug. 2015.
- [76] Z. Bai, M. Li, Y. Dong, H. Zhang, and P. Ma, "Joint fair resource allocation of D2D communication underlaying downlink cellular system with imperfect CSI," *IEEE Access*, vol. 6, pp. 63131–63142, 2018.
- [77] R. Chen, X. Liao, S. Zhu, and Z. Liang, "Capacity analysis of device-todevice resource reusing modes for cellular networks," in *Proc. IEEE Int. Conf. Commun., Netw. Satell. (ComNetSat)*, Jul. 2012, pp. 64–68.
- [78] F. Idris, J. Tang, and D. K. C. So, "Resource and energy efficient device to device communications in downlink cellular system," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2018, pp. 1–6.
- [79] M. Ali, S. Qaisar, M. Naeem, and S. Mumtaz, "Energy efficient resource allocation in D2D-assisted heterogeneous networks with relays," *IEEE Access*, vol. 4, pp. 4902–4911, 2016.

- [80] A. Shahid, K. S. Kim, E. De Poorter, and I. Moerman, "Self-organized energy-efficient cross-layer optimization for device to device communication in heterogeneous cellular networks," *IEEE Access*, vol. 5, pp. 1117–1128, 2017.
- [81] Y. Huang, A. A. Nasir, S. Durrani, and X. Zhou, "Mode selection, resource allocation, and power control for D2D-enabled two-tier cellular network," *IEEE Trans. Commun.*, vol. 64, no. 8, pp. 3534–3547, Aug. 2016.
- [82] M. Zulhasnine, C. Huang, and A. Srinivasan, "Efficient resource allocation for device-to-device communication underlaying LTE network," in *Proc. IEEE 6th Int. Conf. Wireless Mobile Comput., Netw. Commun.*, Oct. 2010, pp. 368–375.
- [83] F. Hussain, M. Y. Hassan, M. S. Hossen, and S. Choudhury, "An optimal resource allocation algorithm for D2D communication underlaying cellular networks," in *Proc. 14th IEEE Annu. Consum. Commun. Netw. Conf.* (CCNC), Jan. 2017, pp. 867–872.
- [84] G. Katsinis, E. Tsiropoulou, and S. Papavassiliou, "Multicell interference management in device to device underlay cellular networks," *Future Internet*, vol. 9, no. 3, p. 44, Aug. 2017.
- [85] P. Phunchongharn, E. Hossain, and D. I. Kim, "Resource allocation for device-to-device communications underlaying LTE-advanced networks," *IEEE Wireless Commun.*, vol. 20, no. 4, pp. 91–100, Aug. 2013.
- [86] D. Wu and N. Ansari, "High capacity spectrum allocation for multiple D2D users reusing downlink spectrum in LTE," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2018, pp. 1–6.
- [87] A. Kose and B. Ozbek, "Resource allocation for underlaying device-todevice communications using maximal independent sets and knapsack algorithm," in *Proc. IEEE 29th Annu. Int. Symp. Pers., Indoor Mobile Radio Commun. (PIMRC)*, Sep. 2018, pp. 1–5.
- [88] Z. Kuang, G. Liu, G. Li, and X. Deng, "Energy efficient resource allocation algorithm in energy harvesting-based D2D heterogeneous networks," *IEEE Internet Things J.*, vol. 6, no. 1, pp. 557–567, Feb. 2019.
- [89] F. W. Zaki, S. Kishk, and N. H. Almofari, "Distributed resource allocation for D2D communication networks using auction," in *Proc. 34th Nat. Radio Sci. Conf. (NRSC)*, Mar. 2017, pp. 284–293.
- [90] H.-H. Nguyen, M. Hasegawa, and W.-J. Hwang, "Distributed resource allocation for D2D communications underlay cellular networks," *IEEE Commun. Lett.*, vol. 20, no. 5, pp. 942–945, May 2016.
- [91] S. Dominic and L. Jacob, "Distributed resource allocation for D2D communications underlaying cellular networks in time-varying environment," *IEEE Commun. Lett.*, vol. 22, no. 2, pp. 388–391, Feb. 2018.
- [92] S. Jung and J. Kim, "A new way of extending network coverage: Relay-assisted D2D communications in 3GPP," *ICT Exp.*, vol. 2, no. 3, pp. 117–121, 2016.
- [93] H. Min, J. Lee, S. Park, and D. Hong, "Capacity enhancement using an interference limited area for device-to-device uplink underlaying cellular networks," *IEEE Trans. Wireless Commun.*, vol. 10, no. 12, pp. 3995–4000, Dec. 2011.
- [94] N. Zlatanov and R. Schober, "Buffer-aided relaying with adaptive link selection—Fixed and mixed rate transmission," *IEEE Trans. Inf. Theory*, vol. 59, no. 5, pp. 2816–2840, May 2013.
- [95] M. Nitti, G. A. Stelea, V. Popescu, and M. Fadda, "When social networks meet D2D communications: A survey," *Sensors*, vol. 19, no. 2, p. 396, 2019.
- [96] Q. Du, M. Liu, Q. Xu, H. Song, L. Sun, and P. Ren, "Interferenceconstrained routing over P2P-share enabled multi-hop D2D networks," *Peer-to-Peer Netw. Appl.*, vol. 10, no. 6, pp. 1354–1370, 2017.
- [97] L. Wang, L. Liu, X. Cao, X. Tian, and Y. Cheng, "Sociality-aware resource allocation for device-to-device communications in cellular networks," *IET Commun.*, vol. 9, no. 3, pp. 342–349, 2015.
- [98] Z. Zhou, C. Zhang, J. Wang, B. Gu, S. Mumtaz, J. Rodriguez, and X. Zhao, "Energy-efficient resource allocation for energy harvestingbased cognitive machine-to-machine communications," *IEEE Trans. Cognit. Commun. Netw.*, vol. 5, no. 3, pp. 595–607, Sep. 2019.
- [99] X.-L. Huang, X. Ma, and F. Hu, "Machine learning and intelligent communications," *Mobile Netw. Appl.*, vol. 23, no. 1, pp. 68–70, 2018.
- [100] M. Meng, Z. Yujun, Z. Dadong, and C. Fan, "Scheduling for data transmission in multi-hop IEEE 802.15. 4E TSCH networks," *Mobile Netw. Appl.*, vol. 23, no. 1, pp. 119–125, 2018.
- [101] Z. Li, J. Chen, Y. Fu, G. Hu, Z. Pan, and L. Zhang, "Community detection based on regularized semi-nonnegative matrix tri-factorization in signed networks," *Mobile Netw. Appl.*, vol. 23, no. 1, pp. 71–79, Feb. 2018.

- [102] B. Li, X. Ding, H. Yang, G. Liu, and X. Peng, "Physical-layer network coding scheme over asymmetric Rayleigh fading two-way relay channels," *Mobile Netw. Appl.*, vol. 23, no. 1, pp. 80–85, Feb. 2018.
- [103] P. Vamvakas, E. E. Tsiropoulou, and S. Papavassiliou, "Dynamic provider selection & power resource management in competitive wireless communication markets," *Mobile Netw. Appl.*, vol. 23, no. 1, pp. 86–99, Feb. 2018.
- [104] B. Gu, X. Zhang, Z. Lin, and M. Alazab, "Deep multiagent reinforcement-learning-based resource allocation for Internet of controllable things," *IEEE Internet Things J.*, vol. 8, no. 5, pp. 3066–3074, Mar. 2021.
- [105] H. Ye, G. Y. Li, and B.-H.-F. Juang, "Deep reinforcement learning based resource allocation for V2 V communications," *IEEE Trans. Veh. Technol.*, vol. 68, no. 4, pp. 3163–3173, Apr. 2019.
- [106] Y. Luo, Z. Shi, X. Zhou, Q. Liu, and Q. Yi, "Dynamic resource allocations based on Q-learning for D2D communication in cellular networks," in *Proc. 11th Int. Comput. Conf. Wavelet Active Media Technol. Inf. Process.* (ICCWAMTIP), Dec. 2014, pp. 385–388.
- [107] K. Zia, N. Javed, M. N. Sial, S. Ahmed, A. A. Pirzada, and F. Pervez, "A distributed multi-agent RL-based autonomous spectrum allocation scheme in D2D enabled multi-tier HetNets," *IEEE Access*, vol. 7, pp. 6733–6745, 2019.
- [108] H. Yang, X. Xie, and M. Kadoch, "Intelligent resource management based on reinforcement learning for ultra-reliable and low-latency IoV communication networks," *IEEE Trans. Veh. Technol.*, vol. 68, no. 5, pp. 4157–4169, May 2019.
- [109] M. Liu, T. Song, J. Hu, J. Yang, and G. Gui, "Deep learning-inspired message passing algorithm for efficient resource allocation in cognitive radio networks," *IEEE Trans. Veh. Technol.*, vol. 68, no. 1, pp. 641–653, Jan. 2019.
- [110] Y. Zhou, Z. M. Fadlullah, B. Mao, and N. Kato, "A deep-learning-based radio resource assignment technique for 5G ultra dense networks," *IEEE Netw.*, vol. 32, no. 6, pp. 28–34, Nov. 2018.
- [111] F. Tang, Z. M. Fadlullah, B. Mao, and N. Kato, "An intelligent traffic load prediction-based adaptive channel assignment algorithm in SDN-IoT: A deep learning approach," *IEEE Internet Things J.*, vol. 5, no. 6, pp. 5141–5154, Dec. 2018.
- [112] M. Yan, G. Feng, J. Zhou, Y. Sun, and Y.-C. Liang, "Intelligent resource scheduling for 5G radio access network slicing," *IEEE Trans. Veh. Technol.*, vol. 68, no. 8, pp. 7691–7703, Aug. 2019.
- [113] A. Asheralieva and Y. Miyanaga, "An autonomous learning-based algorithm for joint channel and power level selection by D2D pairs in heterogeneous cellular networks," *IEEE Trans. Commun.*, vol. 64, no. 9, pp. 3996–4012, Sep. 2016.
- [114] L. Liang, H. Ye, and G. Y. Li, "Spectrum sharing in vehicular networks based on multi-agent reinforcement learning," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 10, pp. 2282–2292, Oct. 2019.
- [115] A. Asheralieva and Y. Miyanaga, "Multi-agent Q-learning for autonomous D2D communication," in *Proc. Int. Symp. Intell. Signal Process. Commun. Syst. (ISPACS)*, Oct. 2016, pp. 1–6.
- [116] K. Zia, N. Javed, M. N. Sial, S. Ahmed, and F. Pervez, "Multi-agent RL based user-centric spectrum allocation scheme in D2D enabled hetnets," in *Proc. IEEE 23rd Int. Workshop Comput. Aided Modeling Design Commun. Links Netw. (CAMAD)*, Sep. 2018, pp. 1–6.
- [117] Z. Li and C. Guo, "Multi-agent deep reinforcement learning based spectrum allocation for D2D underlay communications," *IEEE Trans. Veh. Technol.*, vol. 69, no. 2, pp. 1828–1840, Feb. 2020.
- [118] S. A. Kazmi, L. U. Khan, N. H. Tran, and C. S. Hong, *Network Slicing for 5G and Beyond Networks*. Cham, Switzerland: Springer, 2019.
- [119] M. Condoluci, F. Sardis, and T. Mahmoodi, "Softwarization and virtualization in 5G networks for smart cities," in *International Internet of Things Summit.* Springer, 2015, pp. 179–186.
- [120] A. A. Barakabitze, I.-H. Mkwawa, L. Sun, and E. Ifeachor, "QualitySDN: Improving video quality using MPTCP and segment routing in SDN/NFV," in *Proc. 4th IEEE Conf. Netw. Softwarization Workshops* (*NetSoft*), Jun. 2018, pp. 182–186.
- [121] X. Zhou, R. Li, T. Chen, and H. Zhang, "Network slicing as a service: Enabling enterprises' own software-defined cellular networks," *IEEE Commun. Mag.*, vol. 54, no. 7, pp. 146–153, Jul. 2016.
- [122] A. Khan, A. Zugenmaier, D. Jurca, and W. Kellerer, "Network virtualization: A hypervisor for the Internet?" *IEEE Commun. Mag.*, vol. 50, no. 1, pp. 136–143, Jan. 2012.

- [123] R. Ravindran, A. Chakraborti, S. O. Amin, A. Azgin, and G. Wang, "5G-ICN: Delivering ICN services over 5G using network slicing," *IEEE Commun. Mag.*, vol. 55, no. 5, pp. 101–107, May 2017.
- [124] L. U. Khan, I. Yaqoob, N. H. Tran, Z. Han, and C. S. Hong, "Network slicing: Recent advances, taxonomy, requirements, and open research challenges," *IEEE Access*, vol. 8, pp. 36009–36028, 2020.
- [125] P. Caballero, A. Banchs, G. De Veciana, and X. Costa-Perez, "Network slicing games: Enabling customization in multi-tenant mobile networks," *IEEE/ACM Trans. Netw.*, vol. 27, no. 2, pp. 662–675, Apr. 2019.
- [126] W. Guan, X. Wen, L. Wang, Z. Lu, and Y. Shen, "A service-oriented deployment policy of end-to-end network slicing based on complex network theory," *IEEE Access*, vol. 6, pp. 19691–19701, 2018.
- [127] D. Bega, M. Gramaglia, A. Banchs, V. Sciancalepore, K. Samdanis, and X. Costa-Perez, "Optimising 5G infrastructure markets: The business of network slicing," in *Proc. IEEE Conf. Comput. Commun. (INFOCOM)*, May 2017, pp. 1–9.
- [128] M. Bagaa, T. Taleb, A. Laghrissi, A. Ksentini, and H. Flinck, "Coalitional game for the creation of efficient virtual core network slices in 5G mobile systems," *IEEE J. Sel. Areas Commun.*, vol. 36, no. 3, pp. 469–484, Mar. 2018.
- [129] Y. L. Lee, J. Loo, T. C. Chuah, and L.-C. Wang, "Dynamic network slicing for multitenant heterogeneous cloud radio access networks," *IEEE Trans. Wireless Commun.*, vol. 17, no. 4, pp. 2146–2161, Apr. 2018.
- [130] A. Laghrissi, T. Taleb, and M. Bagaa, "Conformal mapping for optimal network slice planning based on canonical domains," *IEEE J. Sel. Areas Commun.*, vol. 36, no. 3, pp. 519–528, Mar. 2018.
- [131] V. Sciancalepore, K. Samdanis, X. Costa-Perez, D. Bega, M. Gramaglia, and A. Banchs, "Mobile traffic forecasting for maximizing 5G network slicing resource utilization," in *Proc. IEEE Conf. Comput. Commun.* (INFOCOM), May 2017, pp. 1–9.
- [132] P. Zhao, H. Tian, S. Fan, and A. Paulraj, "Information prediction and dynamic programming-based RAN slicing for mobile edge computing," *IEEE Wireless Commun. Lett.*, vol. 7, no. 4, pp. 614–617, Aug. 2018.
- [133] A. Baumgartner, T. Bauschert, A. M. C. A. Koster, and V. S. Reddy, "Optimisation models for robust and survivable network slice design: A comparative analysis," in *Proc. IEEE Global Commun. Conf. (GLOBE-COM)*, Dec. 2017, pp. 1–7.
- [134] R. Wen, G. Feng, W. Tan, R. Ni, S. Qin, and G. Wang, "Protocol function block mapping of software defined protocol for 5G mobile networks," *IEEE Trans. Mobile Comput.*, vol. 17, no. 7, pp. 1651–1665, Jul. 2018.
- [135] Z. Zhou, B. Wang, B. Gu, B. Ai, S. Mumtaz, J. Rodriguez, and M. Guizani, "Time-dependent pricing for bandwidth slicing under information asymmetry and price discrimination," *IEEE Trans. Commun.*, vol. 68, no. 11, pp. 6975–6989, Nov. 2020.
- [136] C. Vlachos and V. Friderikos, "Optimal virtualized resource slicing for device-to-device communications," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2014, pp. 1–7.
- [137] Y. Cai, F. R. Yu, and C. Liang, "Resource sharing for software defined D2D communications in virtual wireless networks with imperfect NSI," in *Proc. IEEE Global Commun. Conf.*, Dec. 2014, pp. 4448–4453.
- [138] B. Cho, K. Koufos, R. Jantti, Z. Li, and M. A. Uusitalo, "Spectrum allocation for multi-operator device-to-device communication," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2015, pp. 5454–5459.
- [139] C. Vlachos, V. Friderikos, and M. Dohler, "Optimal virtualized intertenant resource sharing for device-to-device communications in 5G networks," *Mobile Netw. Appl.*, vol. 22, no. 6, pp. 1010–1019, Dec. 2017.
- [140] A. Moubayed, A. Shami, and H. Lutfiyya, "Wireless resource virtualization with device-to-device communication underlaying LTE network," *IEEE Trans. Broadcast.*, vol. 61, no. 4, pp. 734–740, Dec. 2015.
- [141] Y. Cheng, L. Yang, and H. Zhu, "Operator profit-aware wireless virtualization for device-to-device communications underlaying LTE networks," *IEEE Access*, vol. 5, pp. 11668–11676, 2017.
- [142] K. Wang, F. R. Yu, H. Li, and Z. Li, "Information-centric wireless networks with virtualization and D2D communications," *IEEE Wireless Commun.*, vol. 24, no. 3, pp. 104–111, Jun. 2017.
- [143] G. Sun, G. O. Boateng, D. Ayepah-Mensah, G. Liu, and J. Wei, "Autonomous resource slicing for virtualized vehicular networks with D2D communications based on deep reinforcement learning," *IEEE Syst. J.*, vol. 14, no. 4, pp. 4694–4705, Dec. 2020.
- [144] S. Abdelwahab, B. Hamdaoui, M. Guizani, and T. Znati, "Network function virtualization in 5G," *IEEE Commun. Mag.*, vol. 54, no. 4, pp. 84–91, Apr. 2016.

- [145] A. Laghrissi and T. Taleb, "A survey on the placement of virtual resources and virtual network functions," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 2, pp. 1409–1434, 2nd Quart., 2019.
- [146] A. Hakiri, A. Gokhale, P. Berthou, D. C. Schmidt, and T. Gayraud, "Software-defined networking: Challenges and research opportunities for future Internet," *Comput. Netw.*, vol. 75, pp. 453–471, Dec. 2014.
- [147] R. Mijumbi, J. Serrat, J.-L. Gorricho, N. Bouten, F. De Turck, and R. Boutaba, "Network function virtualization: State-of-the-art and research challenges," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 1, pp. 236–262, 1st Quart., 2016.
- [148] K. Samdanis, X. Costa-Perez, and V. Sciancalepore, "From network sharing to multi-tenancy: The 5G network slice broker," *IEEE Commun. Mag.*, vol. 54, no. 7, pp. 32–39, Jul. 2016.
- [149] J. Kim, D. Kim, and S. Choi, "3GPP SA2 architecture and functions for 5G mobile communication system," *ICT Exp.*, vol. 3, no. 1, pp. 1–8, Mar. 2017.
- [150] I. Alam, K. Sharif, F. Li, Z. Latif, M. M. Karim, S. Biswas, B. Nour, and Y. Wang, "A survey of network virtualization techniques for Internet of Things using SDN and NFV," ACM Comput. Surv., vol. 53, no. 2, pp. 1–40, Jul. 2020.
- [151] (Dec. 2016). Open Networking Foundation, 'ONFTR-526 Applying SDN Architecture to 5G Slicing'. [Online]. Available: https://opennetworking.org/
- [152] Y. Wang, M. Sheng, X. Wang, L. Wang, and J. Li, "Mobile-edge computing: Partial computation offloading using dynamic voltage scaling," *IEEE Trans. Commun.*, vol. 64, no. 10, pp. 4268–4282, Oct. 2016.
- [153] C. You, K. Huang, and H. Chae, "Energy efficient mobile cloud computing powered by wireless energy transfer," *IEEE J. Sel. Areas Commun.*, vol. 34, no. 5, pp. 1757–1771, May 2016.
- [154] S. Sardellitti, G. Scutari, and S. Barbarossa, "Joint optimization of radio and computational resources for multicell mobile-edge computing," *IEEE Trans. Signal Inf. Process. over Netw.*, vol. 1, no. 2, pp. 89–103, Jun. 2015.
- [155] Y.-H. Kao, B. Krishnamachari, M.-R. Ra, and F. Bai, "Hermes: Latency optimal task assignment for resource-constrained mobile computing," *IEEE Trans. Mobile Comput.*, vol. 16, no. 11, pp. 3056–3069, Nov. 2017.
- [156] J. Ren, G. Yu, Y. Cai, and Y. He, "Latency optimization for resource allocation in mobile-edge computation offloading," *IEEE Trans. Wireless Commun.*, vol. 17, no. 8, pp. 5506–5519, Aug. 2018.
- [157] J. Liu, Y. Mao, J. Zhang, and K. B. Letaief, "Delay-optimal computation task scheduling for mobile-edge computing systems," in *Proc. IEEE Int. Symp. Inf. Theory (ISIT)*, Jul. 2016, pp. 1451–1455.
- [158] Y. Li, L. Sun, and W. Wang, "Exploring device-to-device communication for mobile cloud computing," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2014, pp. 2239–2244.
- [159] W. Hu and G. Cao, "Quality-aware traffic offloading in wireless networks," *IEEE Trans. Mobile Comput.*, vol. 16, no. 11, pp. 3182–3195, Nov. 2017.
- [160] L. Pu, X. Chen, J. Xu, and X. Fu, "D2D fogging: An energy-efficient and incentive-aware task offloading framework via network-assisted D2D collaboration," *IEEE J. Sel. Areas Commun.*, vol. 34, no. 12, pp. 3887–3901, Dec. 2016.
- [161] C. You and K. Huang, "Exploiting non-causal CPU-state information for energy-efficient mobile cooperative computing," *IEEE Trans. Wireless Commun.*, vol. 17, no. 6, pp. 4104–4117, Jun. 2018.
- [162] Y. He, J. Ren, G. Yu, and Y. Cai, "D2D communications meet mobile edge computing for enhanced computation capacity in cellular networks," *IEEE Trans. Wireless Commun.*, vol. 18, no. 3, pp. 1750–1763, Mar. 2019.
- [163] J. Sanchez Vilchez, I. Ben Yahia, N. Crespi, T. Rasheed, and D. Siracusa, "Softwarized 5G networks resiliency with self-healing," in *Proc. 1st Int. Conf. 5G Ubiquitous Connectivity*, 2014, pp. 229–233.
- [164] Y. Liu, J. E. Fieldsend, and G. Min, "A framework of fog computing: Architecture, challenges, and optimization," *IEEE Access*, vol. 5, pp. 25445–25454, 2017.
- [165] S. Redana et al., 5G PPP Architecture Working Group: View on 5G Architecture. Brussels, Belgium: European Commission, 2019.
- [166] A. A. Barakabitze and T. Xiaoheng, "Caching and data routing in information centric networking (ICN): The future Internet perspective," *Int. J.*, vol. 4, no. 11, pp. 8–20, 2014.
- [167] C. Tselios and G. Tsolis, "On QoE-awareness through virtualized probes in 5G networks," in *Proc. IEEE 21st Int. Workshop Comput. Aided Modeling Design Commun. Links Netw. (CAMAD)*, Oct. 2016, pp. 159–164.

- [168] Y. C. Hu, M. Patel, D. Sabella, N. Sprecher, and V. Young, "Mobile edge computing—A key technology towards 5G," *ETSI White Paper*, vol. 11, no. 11, pp. 1–16, 2015.
- [169] T. K. Forde, I. Macaluso, and L. E. Doyle, "Exclusive sharing & virtualization of the cellular network," in *Proc. IEEE Int. Symp. Dyn. Spectr. Access Netw. (DySPAN)*, May 2011, pp. 337–348.
- [170] C. Liang and F. Yu, "Wireless virtualization for next generation mobile cellular networks," *IEEE Wireless Commun.*, vol. 22, no. 1, pp. 61–69, Feb. 2015.
- [171] M. Boban, K. Manolakis, M. Ibrahim, S. Bazzi, and W. Xu, "Design aspects for 5G V2X physical layer," in *Proc. IEEE Conf. Standards Commun. Netw. (CSCN)*, Oct. 2016, pp. 1–7.
- [172] N. Hassan, K.-L.-A. Yau, and C. Wu, "Edge computing in 5G: A review," *IEEE Access*, vol. 7, pp. 127276–127289, 2019.
- [173] L. Bing, Z. Yunyong, and X. Lei, "An mec and nfv integrated network architecture," *ZTE Commun.*, vol. 15, no. 2, pp. 19–25, 2019.
- [174] B. Blanco, J. O. Fajardo, I. Giannoulakis, E. Kafetzakis, S. Peng, J. Pérez-Romero, I. Trajkovska, P. S. Khodashenas, L. Goratti, M. Paolino, E. Sfakianakis, F. Liberal, and G. Xilouris, "Technology pillars in the architecture of future 5G mobile networks: NFV, MEC and SDN," *Comput. Standards Interfaces*, vol. 54, pp. 216–228, Nov. 2017.
- [175] Y. Cai, F. R. Yu, C. Liang, B. Sun, and Q. Yan, "Software-defined device-to-device (D2D) communications in virtual wireless networks with imperfect network state information (NSI)," *IEEE Trans. Veh. Technol.*, vol. 65, no. 9, pp. 7349–7360, Sep. 2016.
- [176] K. Wang, H. Li, F. Richard Yu, and W. Wei, "Virtual resource allocation in software-defined information-centric cellular networks with device-todevice communications and imperfect CSI," *IEEE Trans. Veh. Technol.*, vol. 65, no. 12, pp. 10011–10021, Dec. 2016.
- [177] S. Yi, Z. Hao, Z. Qin, and Q. Li, "Fog computing: Platform and applications," in *Proc. 3rd IEEE Workshop Hot Topics Web Syst. Technol.* (*HotWeb*), Nov. 2015, pp. 73–78.
- [178] X. Lin, J. Andrews, A. Ghosh, and R. Ratasuk, "An overview of 3GPP device-to-device proximity services," *IEEE Commun. Mag.*, vol. 52, no. 4, pp. 40–48, Apr. 2014.
- [179] K. Doppler, C. B. Ribeiro, and J. Kneckt, "Advances in D2D communications: Energy efficient service and device discovery radio," in *Proc. 2nd Int. Conf. Wireless Commun., Veh. Technol., Inf. Theory Aerosp. Electron. Syst. Technol. (Wireless VITAE)*, Feb. 2011, pp. 1–6.
- [180] C. Singhal and S. De, Resource Allocation in Next-Generation Broadband Wireless Access Networks. Hershey, PA, USA: IGI Global, 2017.
- [181] O. Hayat, R. Ngah, and Y. Zahedi, "In-band device to device (D2D) communication and device discovery: A survey," *Wireless Pers. Commun.*, vol. 106, no. 2, pp. 451–472, May 2019.
- [182] E. Gures, I. Shayea, A. Alhammadi, M. Ergen, and H. Mohamad, "A comprehensive survey on mobility management in 5G heterogeneous networks: Architectures, challenges and solutions," *IEEE Access*, vol. 8, pp. 195883–195913, 2020.
- [183] Y. Zhai, T. Bao, L. Zhu, M. Shen, X. Du, and M. Guizani, "Toward reinforcement-learning-based service deployment of 5G mobile edge computing with request-aware scheduling," *IEEE Wireless Commun.*, vol. 27, no. 1, pp. 84–91, Feb. 2020.
- [184] H. Xiang, M. Peng, Y. Sun, and S. Yan, "Mode selection and resource allocation in sliced fog radio access networks: A reinforcement learning approach," *IEEE Trans. Veh. Technol.*, vol. 69, no. 4, pp. 4271–4284, Apr. 2020.
- [185] P. Marsch, I. Da Silva, O. Bulakci, M. Tesanovic, S. E. El Ayoubi, T. Rosowski, A. Kaloxylos, and M. Boldi, "5G radio access network architecture: Design guidelines and key considerations," *IEEE Commun. Mag.*, vol. 54, no. 11, pp. 24–32, Nov. 2016.
- [186] R. Ferrus, O. Sallent, J. P. Romero, and R. Agusti, "On 5G radio access network slicing: Radio interface protocol features and configuration," *IEEE Commun. Mag.*, vol. 56, no. 5, pp. 184–192, May 2018.
- [187] V. Ortega, F. Bouchmal, and J. F. Monserrat, "Trusted 5G vehicular networks: Blockchains and content-centric networking," *IEEE Veh. Technol. Mag.*, vol. 13, no. 2, pp. 121–127, Jun. 2018.
- [188] J. Ni, X. Lin, and X. S. Shen, "Efficient and secure service-oriented authentication supporting network slicing for 5G-enabled IoT," *IEEE J. Sel. Areas Commun.*, vol. 36, no. 3, pp. 644–657, Mar. 2018.
- [189] J. Feng, Q. Pei, F. R. Yu, X. Chu, J. Du, and L. Zhu, "Dynamic network slicing and resource allocation in mobile edge computing systems," *IEEE Trans. Veh. Technol.*, vol. 69, no. 7, pp. 7863–7878, Jul. 2020.



LUBNA NADEEM received the B.S. and M.S. degrees in telecommunication engineering from the University of Engineering and Technology (UET), Taxila, Pakistan, in 2011 and 2013, respectively. She is currently with the Communication Systems Research Group, Queen Marry University of London, U.K., and the ACTSENA Research Group, UET, where she is also a Lecturer with the Department of Telecommunication Engineering. Her main research interests include 5G wireless

communication, radio resource allocation and optimization in cellular networks, Internet of Things with machine learning, and deep learning.



MUHAMMAD AWAIS AZAM received the B.Sc. degree (Hons.) in computer engineering from the University of Engineering and Technology (UET), Taxila, Pakistan, in 2006, the M.Sc. degree (Hons.) in wireless networks from Queen Mary University, London, U.K., in 2008, and the Ph.D. degree in pervasive and ubiquitous computing from London in 2012. From 2006 to 2007, he was a Lecturer with UET. From 2012 to 2013, he was the Head of academics with the Cromwell

College of IT and Management, London. From 2013 to 2019, he was an Assistant Professor with the Department of Computer Engineering, UET. He is currently the Head of the School of Information Technology, Whitecliffe. He leads a research team of M.Sc. and Ph.D. students in the area of pervasive and ubiquitous computing. His research interest includes network architecture, IoT, network security, embedded systems, wireless communications, opportunistic networks, and recommender systems. He received the Gold Medal for his B.Sc. degree in 2006.



YASAR AMIN (Senior Member, IEEE) received the B.Sc. degree in electrical engineering with specialization in telecommunications, the M.Sc. degree in electrical engineering with specialization in system on chip design, and the Ph.D. degree in electronic and computer systems from the KTH Royal Institute of Technology, Sweden, with a research focus on printable green RFID antennas for embedded sensors. He is currently a Professor and the Chairman with the Department of

Telecommunication Engineering, University of Engineering and Technology, Taxila, Pakistan. He is the founder of Agile Creative Technologies for Smart Electromagnetic Novel Applications Research Group. He has authored or coauthored more than 100 international technical articles in conferences and journals. His research interests include the next-generation mobile communication systems, millimeter-wave and terahertz antenna array, and implantable and wearable electronics in microwave applications. He is a member of more than a dozen international professional societies and a Fellow of PAE.



MOHAMMED A. AL-GHAMDI received the B.Sc. degree in computer science from King Abdulaziz University, Saudi Arabia, in 2006, and the master's and Ph.D. degrees in computer science from the University of Warwick, U.K., in 2008 and 2012, respectively. He is currently an Assistant Professor with Umm Al-Qura University, Saudi Arabia. He has also authored or coauthored a number of good quality journal articles in signal and image processing domain and various conference papers. His research interests include wireless networks and

4G/5G networks



KOK KEONG CHAI (MICHAEL) received the B.Eng. (Hons), M.Sc., and Ph.D. degrees in 1998, 1999, and 2007, respectively. He joined the School of Electronic Engineering and Computer Science, Queen Mary University of London (QMUL), in 2008. He is currently a Reader of the Internet of Things and the Queen Mary Co-Director of Joint Program with the Beijing University of Posts and Telecommunications, and a member of the Communication Systems Research Group,

QMUL. He has authored more than 65 technical journals and conference papers in the areas of machine to machine communications, wireless communications, the Internet of Things, and smart grids. His current research interests include sensing and prediction in distributed smart grid networks, smart energy charging schemes, applied blockchain technologies, dynamic resource management wireless communications, and medium access control for machine to machine communications and networks.



MUHAMMAD FAISAL NADEEM KHAN received the B.Sc. degree in electrical engineering from Air University, Islamabad, Pakistan, and the M.Sc. degree in electrical engineering from the University of Engineering and Technology, Taxila, Pakistan, and the Ph.D. degree in electrical engineering from the University of Engineering and Technology, Taxila, in 2019. He is currently an Associate Professor with the Department of Electrical Engineering, University of Engineering

and Technology, Taxila, Pakistan. He has authored or coauthored more than 40 refereed journal and conference papers in the smart energy systems related areas. His research interests include power system planning, renewable energy modeling, load modeling, and smart grids.



MUHAMMAD ADNAN KHAN received the B.S. and M.Phil. degrees from the International Islamic University, Islamabad, Pakistan, and the Ph.D. degree from ISRA University, Pakistan. He has worked in various academic and industrial roles in Pakistan. He is currently an Associate Professor with the Faculty of Computing, Riphah School of Computing & Innovation, Riphah International University, Lahore Campus, Lahore, Pakistan. He has been teaching grad-

uate and undergraduate students in computer science and engineering for the past 12 years. He is currently guiding five Ph.D. scholars and six M.Phil. scholars. His B.S. and M.Phil. degrees were supported by scholarship award from the Punjab Information and Technology Board, Government of Punjab, Pakistan. His Ph.D. degree was supported by a scholarship award from the Higher Education Commission, Islamabad, Pakistan. He has authored or coauthored more than 175 research articles with more than cumulative JCR-IF 200 in international journals and reputed international conferences. His research interests primarily include machine learning, MUD, image processing and medical diagnosis, and channel estimation in multi-carrier communication systems using soft computing, with various publications in journals having more than 200 cumulative JCR-IF and conferences of international repute.